UNIVERSITY OF THE WESTERN CAPE

THE RELATIONSHIP BETWEEN FAMILY RESILIENCE AND ACADEMIC PERFORMANCE OF LEARNERS IN THE PHASE OF MIDDLE CHILDHOOD

Esther Alard

Student no: 8300520

Full thesis submitted in fulfilment of the requirements for the degree MA (CFS) in the Department of Social Work, Faculty of Community and Health Sciences, University of the Western Cape

January 2016

Supervisor: Professor N. Roman

Co-Supervisor: Ms S. Isaacs

ABSTRACT

Family resilience has a positive bearing on the academic performance of learners in the developmental phase of middle childhood. The role of family is often ignored and so there is a gap in the literature on the link between positive academic performance and family resilience. The aim of the study was to examine the relationship between perceived family resilience and the academic performance of children in the phase of middle childhood. A quantitative methodological approach was employed in this study with a cross-sectional correlational design. The type of sampling used in this study was convenience sampling. Three primary schools were selected and participants at the schools were randomly selected. The sample consisted of N = 194 Grade 6 learners from schools in the Penlyn Estate area. The reason for having chosen Grade 6 learners was that they are on the brink of puberty and have a good idea of how things function within the family and they were therefore better able to verbalise their opinions than were the younger learners in the phase of middle childhood. The data was collected using a self-reported questionnaire that included the demographic information and the Family Resilience Assessment Scale, as part of the quantitative methodology. The data was then analysed using the Statistical Package for Social Sciences V23 (SPSS). The results were provided using descriptive and inferential statistics. Participation in this study was voluntary after being well informed, while confidentiality and anonymity were maintained throughout the study. The results show that there is a significant positive relationship between the dimensions of academic performance and family resilience.

DEFINITIONS OF TERMS

Families: a societal group that is related by blood (kinship), adoption, foster care or the ties of marriage (civil, customary or religious), civil union or cohabitation, and go beyond a particular physical residence. Department of Social Development (2012).

Resilience: The possession of several skills, in varying degrees, that help a person cope (Alvord & Grados, 2005). The common thread is that people have been able to lead more successful lives than expected despite being at greater risk than average for serious problems (Brooks, 2006).

Family resilience: This term refers to the family's ability to actively "bounce back" after experiencing a crisis or challenge, strengthened and more resourceful to meet the challenges of life. Families who are strengthened and more resourceful after a crisis or a challenge are not simply surviving or managing but growing and thriving (Walsh, 2002). The total scale score from the Family Resilience Assessment Scale (FRAS) measured family resilience.

Academic performance: The multidimensional characteristics of a learner – including skills, attitudes, and behaviours – that factor into their academic success. These characteristics can be separated and considered in one of two primary domains: academic skills or academic enablers (DiPerna & Elliot, 2000; Elliot & DiPerna, 2002). Academic skills are both the basic and complex skills (e.g., reading, writing, calculating, and critical thinking) needed to access and interact with content-specific knowledge. Academic enablers, however, are the attitudes and behaviours (e.g., interpersonal skills, motivation, study skills, and engagement) that a learner needs in order to take advantage of education.

Middle childhood: A distinct developmental stage between early childhood and adolescence, defined by increasing cognitive development, emotional regulation, and relative social independence. (Campbell, 2011)

Family communication patterns: Family communication refers to the way verbal and non-verbal information is exchanged between family members (Epstein, Bishop, Ryan, Miller, & Keitner, 1993).

Belief systems: a combination of feelings, thoughts, actions, and expectations (Sykes, 1989).

Family organisational patterns: Family resilience is influenced by flexibility within the family, cohesion, and social and economic resources that interact to make the patterns of functioning. Family patterns lead to either healthy functioning or unhealthy functioning (Walsh, 2003a).

Education: teaching, instruction or training provided to *learners* (Western Cape Provincial School Education, 1977)

Family resilience framework: A family resilience framework was developed (Walsh, 1996, 1998b) to guide clinical practice. This framework is informed by research in the social sciences and clinical practice seeking to understand crucial variables contributing to individual resilience and well-functioning families (Walsh 1996, 1998b). By definition a family resilience framework focuses on strengths under stress, in the midst of crisis and overwhelming adversity (Walsh, 2003c)

DECLARATION

I declare that The relationship between family resilience and academic performance of learners in the phase of middle childhood is my own work, that it has not been previously submitted for any degree or examination, and that all sources have been acknowledged.



Esther Alard

Signed:....

ACKNOWLEDGEMENTS

I wish to thank God, my heavenly Father, for the grace and provision that has brought me thus far. Many people have helped me on this journey, and to all of them I'd like to give recognition:

To my immediate family I wish to express my heartfelt thanks for their unwavering love, support and patience throughout the journey. My husband, Rubin Alard's vote of confidence meant much to me. There were times that I felt like throwing in the towel, and my daughter, Zenith, had the words of encouragement I needed to persevere. My son, Ezra, constantly asked me how far I was, and what the next step would be. Being a sports enthusiast and in his attempts to keep me going (especially towards the end), he likened the end to the last 100m of a long-distance race. I got the picture, and his words kept me pushing onward. My grand-daughter, Kenzie, was considerate during times when I couldn't give her attention or do things with her, and she would occupy herself in my room, whilst I was busy working.

To my siblings, in-laws and extended family (including my church family), I would like to express huge appreciation for their always enquiring as to how things were taking shape and for moral support.

To my principal, Mrs Sukanya Naidoo, who has been gracious in allowing me time off to meet with other school principals, to do data collection, and to leave school early for meetings with my supervisors, I also wish to express gratitude. To another colleague who has made a meaningful contribution to the presentation of my completed work and also in terms of assisting me with technology, Sean Pillay, I also am very appreciative.

My supervisor, Professor Roman, has been an amazing source of support, in guiding and advising me in a professional capacity. Her years of experience and insight into community

health sciences, have provided meaningful insight and strategy. She had a way of bringing balance to the situation when I was feeling overwhelmed.

Huge thanks goes to my co-supervisor, Ms Serena Isaacs, for her input, feedback and encouragement throughout the study. She has taught me much from the start to completion of the study.

I have appreciated the social media (Watsapp) group interaction and motivation provided by my fellow Masters' students, many of whom were available to offer helpful and needed advice.

Final thanks goes to the CHS administrator, Arthur Chikware, for his administration and assistance throughout the course.



TABLE OF CONTENTS

CHA	APTER 1	14
INT	RODUCTION	14
1.1	Background	14
1.2	Middle childhood	14
1.2	2.1 Academic achievemen	nt in middle childhood16
1.2	2.2 The role of families in	a academic achievement in middle childhood18
1.3	Family resilience	19
1.4	Problem statement	22
1.5	Research questions, aim	and objectives23
1.5	5.1 Research question	UNIVERSITY of the WESTERN CAPE
1.5	The aim of the study.	23
1.5	5.3 Objectives	24
1.6	Hypotheses	24
1.7	Significance of the study	y24
1.8	Outline of thesis	25
CHA	APTER 2	27
THE	CORETICAL FRAMEWO	ORK27
2.1	Introduction	27
2.2	Resilience	27

2.3 H	listory of the concept of resilience	28
2.4 T	The concept of family resilience	30
2.5 Fa	amily resilience framework	32
2.6 K	Keys to resilience	34
2.6.1	Family belief systems	34
2.6.2	Family organization	35
2.6.3	Communication	35
2.7 P	rominent family resilience factors	36
2.7.1	Positive outlook	
2.7.2	Family member accord	37
2.7.3	Financial management	38
2.7.4	Family time	38
2.7.5	Shared recreation	39
2.7.6	Routines and Rituals	41
2.7.7	Support network	41
2.8 L	earner academic performance	43
2.8.1	Course grades	45
	amily resilience and children's learning	
2.10 C	Conclusion	47

CHA	PTE	ER 3	48
LITI	E RA I	TURE REVIEW	48
3.1	Int	roduction	48
3.2	Mic	ddle childhood	48
3.2	2.1	Cognitive functioning	49
3.2	2.2	Social functioning	49
3.2	2.3	Emotional functioning	51
3.3	Ass	sumptions of family life	52
3.4	Far	mily structure and well-being	53
3.4	.1	Family processes	55
3.4	1.2	Positive and involved parenting	58
3.4	1.3	WESTERN CAPE Effective discipline	59
3.4	.4	Healthy relationships	59
3.5	Far	mily resilience studies	60
3.5	5.1	Longitudinal study in children's resilience	62
3.5	5.2	Strength within families	63
3.6	Aca	ademic performance	63
3.6	5.1	Predictors of academic performance	64
3.6	5.2	Importance of academic performance	65

3.6.3 How family influences academic performance	65
3.7 Conclusion	67
CHAPTER 4	68
METHODOLOGY	68
4.1 Introduction	68
4.2 Aim and objectives of the study	68
4.3 Methodological approach	69
4.4 Research design	69
4.5 Population and sample	70
4.6 Data collection instrument	71
4.6.1. The Family Resilience Assessment Scale	71
4.6.2 Academic performance measurement	71
4.7 Data collection	72
4.7.1 Pilot study	73
4.7.1.1 Challenges identified during the pilot study	74
4.8 Data collection for the main study	76
4.9 Data analysis	76
4.10 Reliability and validity	77
4.11 Ethics statement	78
4.12 Conclusion	79

CHA	PTER 58	0
RESU	JLTS8	0
5.1	Introduction8	0
5.2	Overview8	1
5.2.	1 Aim of the study8	1
5.2.	2 Objectives8	1
5.3	Internal consistency8	1
5.4	Perception of family resilience	4
5.5	Academic Achievement9	1
5.6	The relationship between the variables9	2
5.7	Summary of findings	4
CHA	PTER 69	6
DISC	USSION, CONCLUSION AND RECOMMENDATIONS9	6
6.1	Introduction9	6
6.2	Prevalence of family resilience in accordance with the subscales of The Famil	y
Resili	ence Assessment Scale9	6
6.2.	Family Communication and Problem Solving (FCPS):9	8
6.2.	2 Utilizing Social and Economic Resources (USER)9	8
6.2.	Maintaining a Positive Outlook (MPO)10	0
6.2.	4 Family Connectedness (FC)10	1

6.2.5	Family Spirituality (FS)101
6.2.6	Ability to Make Meaning of Adversity (AMMA)102
6.3 The	e relationship between family resilience and academic performance of learners
in the pha	se of middle childhood103
6.3.1	Maintaining a Positive Outlook (MPO)105
6.3.2	Family connectedness (FC)105
6.3.3	Ability to Make Meaning of Adversity (AMMA)106
6.3.4	Family Communication and Problem Solving (FCPS)106
6.3.5	Utilising Social and Economic Resources (USER)107
6.3.6	Family spirituality107
6.4 Hov	w children may compensate for difficulties they experience in the academic
domain	WESTERN CAPE 109
6.5 Lin	nitations110
6.6 Rec	commendations110
6.7 Con	nclusion112
APPEND	IX Ii
APPEND	IX IIiv
APPEND	IX IIIvii
APPEND	IX IVx
APPENDI	IX Vxiii

APPENDIX VI	XV
APPENDIX VII	xvii

LIST OF TABLES

Table 4.1	Changes made to the measuring instrument
Table 5.1	Internal consistency table
Table5.2	Demographic information of participants
Table 5.3	Subscale on Family Communication and Problem Solving (FCPS)
Table 5.4	Utilizing Social and Economic Resources (USER)
Table 5.5	Maintaining a Positive Outlook (MPO)
Table 5.6	Family Connectedness (FC)
Table 5.7	Family Spirituality (FS)
Table 5.8	Ability to Make Meaning of Adversity (AMMA
Table 5.9	Table of the mean scores for the variables of family resilience
Table 5.10	Percentage scores for the academic achievement for the total sample (N=194)
Table 5.11	The relationship between family resilience and academic performance
Table 5.12	Regression analyses predicting academic performance

CHAPTER 1

INTRODUCTION

1.1 Background

What qualities embody or define 'strength' in a family? Indeed, what qualities define a family? How can the work of human services professionals enhance, strengthen and support families' lives? Researchers and practitioners alike grapple with such questions. The answers we provide, today and in the future, will shape and direct educational efforts, preventive interventions, and treatment approaches for children and their families into the 21st century. Politicians and media sources announce that the family is in jeopardy from multiple stressors at institutional and individual levels. Yet parents continue to nurture the children in their care and want the best for them, as parents always have; and families continue to overcome odds, thrive, evolve, and inspire (Gillespie & Primaveray, 2000).

WESTERN CAPE

1.2 Middle childhood

Middle childhood is a crucial yet under-appreciated phase of human development (Del Giudice, 2014). According to Cherry (2014) children in middle childhood experience many developmental changes, both physically and emotionally. Middle childhood is marked by the continued development of cognitive skills, particularly those related to the ability to reason. During this time, trajectories of achievement are established and thereafter remain relatively stable (Kowaleski-Jones & Duncan, 1999).

A crucial shift in children's cognitive skills occurs at around age six. Although the cognitive changes that occur during infancy and the preschool years are dramatic (as children learn their native language, for instance), almost all theories of development point to age six as the time

when children begin to actually reason in the common-sense meaning of the word. All cultures that provide formal schooling for their children begin between ages five and seven (Sameroff, 1996). Although the origin of the change is not well understood, there is a broad consensus that children develop key thinking or conceptual skills during this transition period, which are then refined and consolidated throughout the middle childhood years.

Middle childhood is marked by several types of advances in learning and understanding. During this period, in school and wherever they spend time, children acquire the fundamental skills considered to be important by their culture, such as reading and arithmetic (Selman, 1980). Skills of self-awareness also develop dramatically in middle childhood. For instance, children develop a notion of how one goes about learning, and they discover that strategies such as studying and practicing can improve learning and performance. They become more able to retrieve information and use it to solve new problems or cope with new situations. Both of these skills require the ability to reflect on what one is doing and what one wants to accomplish, and that ability increases dramatically during middle childhood. Children begin to plan consciously, coordinate actions, evaluate their progress, and modify their plans and strategies based on reflection and evaluation (Eccles, 1999). Finally, alongside their increasing ability to reflect on themselves, children also develop the ability to take the perspective of others. They come to understand that others have a different point of view and different knowledge than they have, and they come to understand that these differences have consequences for their interactions with other people. Through their growing understanding of other people's behaviour and through their grasp of written materials, children take in information that builds their knowledge base and stretches their reasoning capacities. The basic mental capacity for all of these skills is in place at a very young age, but it is during middle childhood that these abilities become salient and conscious (Selman, 1980).

1.2.1 Academic achievement in middle childhood

The development of good academic skills is a critical milestone of successful negotiation of the middle childhood period (Alexander, Entwisle & Horsey, 1997). A number of factors contribute to a learner's academic achievement. Among other factors, individual differences in traits, self-esteem, and self-efficacy beliefs have proven to be important predictors of academic achievement (Marsh, Trautwein, Ludtke, Koller, & Baumert, 2006).

Development of good academic skills and the formation of positive peer relationships are critical milestones of successful negotiation of the middle childhood period. Ve'ronneau, Vitaro, Pedersen, and Tremblay (2008) state that academic skills in the elementary school years predict later academic achievement and psychosocial adjustment in adulthood. Academic difficulties may lead to long-term patterns of school drop-out, academic failure, and problems entering a successful career in adulthood (Alexander et al., 2001).

Although the importance of academic achievement is rarely questioned, reaching unanimity regarding its measurement has been elusive. The measurement of academic performance continues to be a controversial topic among policymakers, measurement experts and educators (Ahearn, 2000; Elliott, 1998; Johnson, 2000; Koretz & Hamilton, 1999; McGrew, Vanderwood, Thurlow, & Ysseldyke, 1995). Measuring academic performance can occur at multiple levels and serves multiple purposes. For example, classroom teachers often conduct formative and summative tests to evaluate student mastery of course content and provide grades for students and parents. State tests are designed primarily to measure progress at the school or school district level. In particular, graduation tests are used to determine whether a student has mastered the minimum content and competencies required to receive a high school diploma. Each of these kinds of assessments engenders significant questions related to test

design, types of decisions supported by the results, alternative assessments, and accommodations (Heubert & Hauser, 1999; Minnema, Thurlow, Bielinski, & Scott, 2001).

Although performance on standardized tests receives the greatest attention in discussions of students' academic performance, teachers' evaluations of performance as indicated in course grades represent a common metric of student performance that often is more directly tied to the day-to-day business of teaching and learning than are annual standardized test scores. Grades serve a number of important functions. They communicate to students and parents information about students' mastery of course content. In high school, a passing grade is also the criterion for a course's contributing to accumulated credit for graduation. Finally, grades provide information for consideration in college admissions (Polloway et al., 1994).

Grades are composite measures that account not only for students' content mastery but often for other factors, such as their class participation, attitudes, progress over time, and attendance. Both general and special educators are known to consider these various factors when grading, but to emphasize different factors. For example, special education teachers are less likely than general educators to consider homework or attendance to be important in grading student performance, but are more likely to consider in-class participation to be important (Blackorby, Wagner, Levine, Cameto, & Guzman, 2003). Moreover, substantial variations in grading practices occur across teachers, schools, and school districts. Academic achievement is one of the most important indices of educational evaluation and reflects the socioeconomic development of every country. Academic achievement is assessed through measuring the degree of each student's institutional learning through his/her scores (Seif, 2001). Some students have sufficient mental abilities but cannot accomplish assignments and tests; they may experience academic achievement deficit and face educational challenges.

In this era of globalization and technological revolution, education is considered as a first step for every human activity. It plays a vital role in the development of human capital and is linked with an individual's well-being and opportunities for better living (Battle & Lewis, 2002). It ensures the acquisition of knowledge and skills that enable individuals to increase their productivity and improve their quality of life. This increase in productivity also leads towards new sources of earning which enhances the economic growth of a country (Saxton, 2000). The quality of students' performance remains a top priority for educators. It is meant for making a difference locally, regionally, nationally and globally. Educators, trainers, and researchers have long been interested in exploring variables contributing effectively for quality of performance of learners. These variables are inside and outside school that affect students' quality of academic achievement. These factors may be termed as student factors, family factors, school factors and peer factors (Crosnoe, Johnson & Elder, 2004). Many factors can affect academic achievement. Researchers have pointed to several different factors, which include: family factors such as the role of family and the impact of parenting WESTERN CAPE styles (Samadi, 2007).

1.2.2 The role of families in academic achievement in middle childhood

Han (2008) describes the family environment as the most important factor in learning and development for children and research has generally demonstrated that children raised in dual-parent homes have better school achievement and higher educational attainment than children from single-parent homes. This may be because two-parent homes have greater resources, time, and attention to offer their children. Multiple caretakers also are better able to facilitate academic engagement and performance (Suárez-Orozco et al., 2009).

1.3 Family resilience

In the 1970s, research began to shift its focus to understanding why some persons succeed in spite of adverse circumstances. This change in focus gave rise to studies on what is now referred to as 'resilience', which until recently has predominantly focused on characteristics of the individual. For instance, a review of the evidence shows that since the 1970s many children who were raised in deprived, adverse environments actually went on to become successful and loving individuals (Luthar, Cicchetti, & Becker, 2000). Research into resilience has since expanded to include various age groups and disadvantaged states, such as chronic illness (McCubbin, Balling, Possin, Frierdic, & Bryne, 2002), mental health issues (Jonker & Greef, 2009), poverty and violence (Luthar, 1999) and, catastrophic life events (Clay, 2002). Included in this array is family resilience (Patterson, 2002a). However, this early deficit framework soon gave way to a strength framework, when researchers discovered that some children did well and grew into healthy, adjusted adults despite living in poor environments during childhood (Werner, 1993; Wolin & Wolin, 1993). According to Pretorius (1998), our countless experiences with the world allow us to develop general beliefs about ourselves, and our world. People's evaluations of themselves, their abilities, support resources and their family environment influence their emotions and behaviour during interactions with the environment. Those who perceive themselves, their support and their environment negatively will have serious doubts about their ability to deal with stressful encounters and could consequently, succumb to possible negative psychological effects. Those who perceive themselves, their support and their environment positively will have a greater belief in their ability to manage stressful encounters.

The family's ability to function, according to Walsh (2002), is determined by the level of resilience in the family. Family resilience, as conceptualized by McCubbin and McCubbin

(1996), is the family's ability to utilize behavioural patterns and functional competence to negotiate, and even thrive through hardships and crises. Hawley and DeHann (1996) conceptualize family resilience as a path followed as families adapt and prosper in the face of stress, both in the present and over time. They believe resilient families respond positively to stressful conditions in unique ways, depending on factors such as developmental level, the combination of risk and protective factors, and the family's shared outlook. Several years later, Patterson (2002) also conceptualized family resilience as the adaptive process families utilize to adapt and function competently following exposure to significant adversity or crises. Walsh (2003) has a similar definition, but extends it to include not only a 'bouncing back' after adversity but the ability to 'bounce forward' as well. Walsh (2002) defines family resilience as more than just the usual resilience definition of overcoming adversity, but "the potential for personal and relational transformation and growth that can be forged out of adversity".

The concept of family resilience extends theory and research on family stress, coping, and adaptation (Simon, Murphy, & Smith, 2005). It entails more than managing stressful conditions, shouldering a burden, or surviving an ordeal. It involves the potential for personal and relational transformation and growth that can be forged out of adversity. By tapping into key processes for resilience, families that are struggling can emerge stronger and more resourceful in meeting future challenges. Members may develop new insights and abilities. A crisis can be a wake-up call, heightening attention to important matters. It can become an opportunity for reappraisal of life priorities and pursuits, stimulating greater investment in meaningful relationships. Studies of strong families have found that when family members weathered a crisis together, their relationships were enriched and more loving than they otherwise might have been (Stinnett & DeFrain, 1985).

This family resilience perspective fundamentally alters the deficit-based lens from viewing struggling families as *damaged* and beyond repair to seeing them as *challenged* by life's adversities, with potential for fostering healing and growth in family members.

At a time of widespread concern about the demise of the family, useful conceptual tools are needed as much as techniques to support and strengthen couples and families. The concept of family resilience is a valuable framework to guide research, intervention, and prevention efforts. While some families are shattered by crisis or persistent stresses, others emerge strengthened and more resourceful. A resiliency-based approach aims to identify and fortify key interactional processes that enable families to withstand and rebound from the disruptive challenges they face. A resiliency lens shifts perspective from viewing families as damaged to seeing them as challenged, and it affirms their reparative potential. This approach is founded on the conviction that both individual and family growth can be forged through collaborative efforts in the face of adversity (Walsh, 1996). Children who are raised in a family where issues are constructively dealt with, are better equipped to thrive academically than children who are raised in a non-resilient family environment (Walsh, 2002). Whilst educators and parents have commonly investigated problems with learners' learning skills or methods of academic attainment by looking mainly at learner problems, the role of family is often ignored and so there is a gap in the literature on the link between positive academic achievement and family resilience. Zimmerman and Arunkumar (1994) state that research that focuses on people in a crisis situation and how they differentially adapt is needed to more fully understand the resilience process.

One of the enduring mysteries that confronts those who work with families and children – and those who are concerned with child and family policy – is why some families respond positively to serious threats and challenges to their wellbeing, while others in similar

circumstances do not manage to do so (Walsh, 2003). The concept of resilience has been developed by researchers to denote positive adaptation under adverse circumstances. Hawley and De Hann (1996) describe the family in two contexts: Firstly, and most commonly, the family can serve as a risk factor raising the vulnerability of family members. Some research outlines the kinds of family factors that create risk for family members (for example severe marital conflict, parental mental illness, and so on), while other research has identified factors that help family members be resilient in the face of family dysfunction (for example research on adult children of alcoholics). According to Potter, Mashburn and Grissmer (2013) the family is described in the context of influencing the acquisition of the academic skills. Adversities within families have the potential to do serious harm to family members. However, some children and their families will thrive despite these adversities (Baumrind, 1991). Specialists in the field of family development and family psychology have attributed this difference as being a function of resilience of the family (Patterson, 2002; Walsh, 2003). Secondly, the family can also serve as a protective factor to boost the resilience of the family members. Protective factors include "a good fit between parent and child, maintenance of family rituals, proactive confrontation of problems, minimal conflict in the home during infancy, the absence of divorce during adolescence, and a productive relationship between a child and his or her mother" (Hawley & De Haan, 1996, p. 285). Walsh (1996, p. 263) comments that "few have considered the family as a potential source of resilience that is, as a resource."

1.4 Problem statement

When a family experiences stress, or a life change (positive or negative), the family enters a period of adjustment (Openshaw, 2011). During this adjustment period, the family system as a whole must adapt and change the way in which it functions (Olson & Gorall, 2003).

Openshaw (2011) states that family dynamics such as family functioning and family resilience, can influence multiple areas of an individual's life. One area that will benefit from these family dynamics, (in particular family resilience) is that of academic achievement. According to Carneiro and Heckman (2003) parental capacities, family resources and family environments matter for children's education and long-term wellbeing. The academic success of most students is largely determined by family influences (Pearce & Lin, 2005). Resilience research has traditionally revolved around individual risk and protective factors that enable people to succeed in spite of adverse circumstances (Bhana & Bachoo, 2011). According to Walsh (1996) researchers and clinicians have overlooked the resilience that can be found in families and fostered in couple and family intervention, because of their prior focus being on individual resilience, most often in surviving dysfunctional families. There has been no known research done in South Africa regarding the relationship between family resilience and academic performance of learners in the phase of middle childhood. This proposed research therefore aspires to examine the relationship between family resilience and academic performance of WESTERN CAPE middle childhood learners.

1.5 Research questions, aim and objectives

1.5.1 Research question

What is the nature of the relationship between perceived family resilience and the academic performance of learners in middle childhood?

1.5.2 The aim of the study

The aim of this study was to examine the relationship between perceived family resilience and the academic performance of learners in the phase of middle childhood.

1.5.3 Objectives

The objective of the study was to

- Determine the perception of family resilience of learners in the phase of middle childhood;
- Assess the academic performance of learners in the phase of middle childhood;
- Examine the relationship between and predictive effects of family resilience and academic performance of learners in the phase of middle childhood

1.6 Hypotheses

Hypothesis: There is a significant predictive relationship between family resilience (family communication and problem-solving, maintaining a positive outlook, making meaning of adversity, family spirituality, family connectedness and utilising social and economic resources) and learners' academic performance in the phase of middle childhood.

1.7 Significance of the study

As yet, there has been little work that has specifically tried to implement the ideas that have emerged from the research on the relationship between family resilience and academic achievement of learners in middle childhood. It may offer parents, educators and future researchers a view of understanding the role of family resilience and its effect on academic performance, as well as inform possible interventions to be put in place. Families will know how they can overcome certain barriers and motivate each other by understanding how resilience and the ability to overcome adversity aid the academic performance of middle childhood learners. Educators may incorporate information on this topic into Life Skills

lessons to create awareness of family resilience. Practitioners may benefit from this study and use the opportunities they have with families attending screenings to discuss various concerns and refer where necessary. This study may also help future researchers to relate other variables to the respondents or different respondents to the same variables used.

1.8 Outline of thesis

Chapter one provides an introduction to this study and describes the aims, background and rationale, theoretical framework and problem statement for the study. It highlights the concepts of middle childhood development, academic achievement and family resilience. The objectives of the formulated research problem and hypotheses are introduced. The chapter ends with the significance of the study and a summary of the chapters.

Chapter two provides an overview of the theoretical framework of family resilience as a theory. It provides an overview of the theoretical underpinning of the study in two parts. The first part is aimed at explaining the history of family resilience, and defines the concept of family resilience. The second part is a detailed explanation of family resilience theory. Further it provides an explanation of family resilience factors, academic performance of learners in the phase of middle childhood, and how family resilience facilitates children's learning.

Chapter three provides literature relating to the four variables that form part of this study. These variables include: assumptions of family life, family resilience, academic achievement and middle childhood. The section on assumptions of family life includes family structure and well-being, family processes, positive and involved parenting, effective discipline and healthy relationships. In the section on family resilience studies, a longitudinal study in children's resilience and strength within families are discussed. The final section discusses academic

performance and middle childhood. In essence, this chapter refers to an understanding of family resilience and academic performance of learners in middle childhood.

Chapter four gives an overview of the methodology employed in this research study. It includes a discussion on the design, as well as, the sampling techniques implemented. The Family Resilience Assessment Scale is introduced. The procedures followed for the pilot and the main study are laid out. The data collection process, the analysis of the data and the ethical considerations that were employed are also discussed.

Chapter five provides an analysis of the findings of the quantitative data collected as well as a presentation using tables. Descriptive quantitative results are analysed using the Statistical package in social sciences (SPSS) and these are presented. The statistical presentation reflects the descriptive, correlations and inferential data. The results include information related to the demographics of the study sample.

Chapter six presents a discussion of the study results in detail. It also provides an overall understanding of knowledge of family resilience and student academic performance. The results are interpreted in this chapter and an outline of the limitations and recommendations of the study is given.

CHAPTER 2

THEORETICAL FRAMEWORK

2.1 Introduction

This chapter explores the aspects of the conceptual framework of this study. The purpose of this chapter is to provide an understanding of family resilience and academic performance of learners in the phase of middle childhood. The first section of this chapter looks into the history of family resilience, and defines the concept of family resilience. This is followed by a detailed explanation of family resilience theory, which is the theoretical framework of this study. Further it provides an explanation of family resilience factors, academic performance of learners in the phase of middle childhood, and how family resilience facilitates children's learning.

WESTERN CAPE

2.2 Resilience

Resilience refers generally to the successful adaptation of a system in response to significant challenges. This concept can be applied to any living organism, as well as a family, a community, a workplace, the military as a whole, a computer system, a country, or a global ecosystem. "Successful adaptation," of course, will be defined in different ways, depending on the values, goals, culture, and historical or scientific context of the people making judgments about success. For individual children, both developmental and cultural context play a role in defining good adaptation. Developmental scientists often define resilience with respect to expected achievements for children of different ages or stages of development, sometimes called developmental tasks (McCormack et al., 2011). Some of these expectations are universal, such as learning to walk or talk. Others are more specific to a culture or situation,

such as learning to hunt or to read sacred scriptures in the original language. Families are often judged by how well they promote the health, development, and goals of their members within their culture or society (Goldenberg & Goldenberg, 2011).

2.3 History of the concept of resilience

Positive, strengthening adaptive outcomes in response to adversity have a long and renowned history. Evidence shows that since the 1970s, many children raised in high-risk environments did not mirror their deprived environments, but rather thrived and grew up to become productive and loving individuals (Luthar et al., 2000). Although many deprived or traumatized children became blocked from growth or trapped in a victim position, Rutter (1987) noted that about half of children exposed to high-risk circumstances overcame them, were able to lead loving and productive lives, and went on to raise their own children well. Research on resilience expanded to include various ages and disadvantaged states, such as poverty and violence (Luthar, 1999), maltreatment (Cicchetti & Rogosch, 1997), chronic illness (White et al., 2004), and catastrophic life events (Clay, 2002). Early efforts to describe resilience focused on personal qualities of resilient individuals, such as autonomy and self-esteem (Masten, 2001).

Although social support and family relationships have been studied as predictors of quality of life, limited studies have been conducted on family resilience, or the way in which the family overcomes and grows from adversity or stressful events. The topic of resilienceviz. the positive adaptation from significant adverse events (Walsh, 2003a), has experienced increasing attention over the last three decades (McCubbin et al., 1996). Orbuch et al., (2005) defined resilience as "the ability and competency of individuals or families to exhibit positive consequences given the stress and hardship associated with adverse and distressing situations" (p. 172). Resilience sprang from researching children, who despite terrible life conditions (e.g.,

poverty, abuse), were able to succeed in life (Sixbey, 2005). When studying these children, researchers began to focus on an innate ability within the individual to 'bounce back' (Luthar & Zieglar, 1991). Researchers termed this ability, "resilience." There has been a discussion on the use of the terms "resiliency" vs. "resilience" (Sixbey, 2005). Resiliency is thought to be a state-like trait within the individual, while the term resilience denotes a process (Sixbey, 2005) in which an individual may not be resilient in every situation.

In the beginning, resilience was primarily associated with individual children. The family was viewed as dysfunctional and, most often, the cause of the adverse events. However, through happenstance, researchers began to observe that resilience was not only an individual process, but also a family systems process (Sixbey, 2005). Werner and Smith (1977) and Cicchetti and Rogosch (1997) were the primary researchers on family resilience (Sixbey, 2005). From their research, an evolution toward a more systemic view of resilience was initiated. Family was no longer viewed as damaged, dysfunctional, and problematic; instead, it was viewed as being challenged by life events (Walsh, 2003a). Moreover, McCubbin et al., (1996) expanded the definition of resilience to include family, defining resilience as the positive behavioural patterns and functional competence of both the individuals and the family unit under stressful or adverse circumstances; these positive behavioural patterns and functional competence determine the family's ability to recover by maintaining the family's integrity as a unit while insuring, and where necessary restoring, the well-being of the family members and the family unit (p. 5).

When compared to the development of other theories, the evolution of resilience research has been an organic, inductive process (Richardson, 2002). Rather than emerging from a structured theoretical approach, resilience inquiry has been guided by researchers' observations about human experience. As a result, the development of resilience research has emphasized its

discontinuous nature (Miller, 2011). That is, individuals do not become resilient in a gradual, predictable manner. By definition, resilience necessitates a disruption in life, such as the experience of significant adversity, which in turn facilitates its development. In other words, without adversity, individuals may have the potential for resiliency (i.e., as a trait or personality characteristic) but not resilience, which describes the developmental process and sustained outcome of experiencing and overcoming adversity (Richardson, 2002).

2.4 The concept of family resilience

The concept of family resilience extends the understanding of family functioning to situations of adversity (Walsh, 2006. pg. 399). Walsh (2010) defines resilience as strength in the context of adversity, the ability to withstand and rebound from stressful life challenges, strengthened and more resourceful. Challenges may range from expectable strains of normative life-cycle transitions, such as retirement, divorce or remarriage, to a sudden job loss or untimely death of a key member, or to the prolonged strains of migration or inner-city violence. How a family deals with such challenges is crucial for individual and family recovery.

Family resilience provides a strong connection within families and the practise of it provides a bond which aids with a child's academic development. According to Shonkoff and Phillips (2000) children show significantly better cognitive and language skills, as well as positive social and emotional development, when they are cared for by adults who are attentive to their needs and who interact with them in encouraging and affectionate ways. The absence of such connections early on can harm a child's ability to develop normally. When children have secure attachments early in life, they tend to have better development, social interactions, and academic achievement (Teo, Carlson, Mathieu & Stroufe. 1996).

Resilience, defined as the ability to withstand and rebound from adversity, has become an important concept in mental health theory and research over the past two decades. It involves a dynamic process encompassing positive adaptation within the context of significant adversity (Luthar, Cicchetti, & Becker, 2000). Researchers have found increasing evidence that the same adversity can result in different outcomes. For example, although many lives are shattered by childhood trauma, others emerge from similar high-risk conditions able to live and love well, evident in the finding that most abused children do not become abusive parents (Kaufman & Ziegler, 1987).

To account for these differences, early studies focused on personal traits associated with resilience, or hardiness, reflecting the dominant cultural ethos of the rugged individual. Resilience was viewed as inborn or acquired on one's own, as in the "invulnerable child" thought to be impervious to stress because of inner fortitude or character armour (Anthony & Cohler, 1987). As research extended beyond situations of parental mental illness or maltreatment to multiple adverse conditions (e.g., socioeconomic disadvantages, urban poverty, community violence, chronic illness, and catastrophic life events), resilience came to be viewed in terms of an interplay of risk and protective processes over time, involving individual, family, and larger socio-cultural influences (Werner, 1993).

Notably, emerging studies of resilient individuals remarked on the crucial influence of significant relationships with caring adults and mentors, such as coaches or teachers, who supported the efforts of at risk children, believed in their potential, and encouraged them to make the most of their lives (Walsh, 1996). A family resilience perspective fundamentally alters that deficit-based lens from viewing troubled families as damaged and beyond repair to seeing them as challenged by life's adversities. Rather than rescuing so-called "survivors" from dysfunctional families, this approach engages distressed families with respect and compassion

for their struggles, affirms their reparative potential, and seeks to bring out their best. Research and intervention efforts to foster family resilience aim both to avoid or reduce pathology and dysfunction and to enhance functioning and well-being (Luthar et al., 2000). Such efforts have the potential to benefit all family members as they fortify relational bonds.

2.5 Family resilience framework

A family resilience framework was developed (Walsh, 1996, 1998b) to guide clinical practice. This framework is informed by research in the social sciences and clinical practice seeking to understand crucial variables contributing to individual resilience and well-functioning families (Walsh 1996, 1998b). Essentially a meta-framework, it can be used with a variety of models of intervention. It offers a conceptual map to identify and target key family processes that reduce the risk of dysfunction, buffer stress, and encourage healing and growth from crisis. The framework draws together findings from numerous studies, identifying and synthesizing key processes within three domains of family functioning: family belief systems, organization patterns, and communication processes (Walsh, 1998b).

A family resilience approach builds on developments to strengthen family capacities to master adversity (Walsh, 1996, 1998b). A family resilience framework can serve as a valuable conceptual map to guide prevention and intervention efforts to support and strengthen vulnerable families in crisis. Family resilience involves more than managing stressful conditions, shouldering a burden, or surviving an ordeal. This approach recognizes the potential for personal and relational transformation and growth that can be forged out of adversity. By encouraging key processes for resilience, families can emerge stronger and more resourceful through their shared efforts. A crisis can be a wakeup call, heightening attention to what matters. It can become an opportunity for reappraisal of priorities, stimulating new or renewed investment in meaningful relationships and life pursuits. In fact, families report that

through weathering a crisis together their relationships were enriched and more loving than they might have been otherwise (Stinnett & DeFrain, 1985).

A family resilience framework is especially timely in helping families with unprecedented challenges as the world around them changes at an accelerated pace (Walsh, 2012). Family cultures and structures are becoming increasingly diverse and fluid. Over an extended family life cycle, adults and their children are moving in and out of increasingly complex family configurations, with each transition posing new adaptational challenges. Amid social, economic, and political upheavals worldwide, families are dealing with many losses, disruptions, and uncertainties. Yet, as Lifton (1993) contends, humans are surprisingly resilient. He compares our predicament and response to that of the Greek god Proteus: Just as he was able to change shape in response to crisis, we create new psychological, social, and family configurations, exploring new options and transforming our lives many times over the life course and the generations. Most families show remarkable resilience in creatively reweaving their family life. Yet stressful transitions and attempts to navigate uncharted territory can contribute to individual and relational distress. A resilience-oriented practice approach assesses individual, couple, and family distress in relation to this larger societal and global context. Families may need help to grieve their actual and symbolic losses as they "bounce forward" to adapt. Therapists can help families to find coherence in the midst of complexity, and maintain continuities in the midst of upheaval as they journey into the future. Resilience does not mean bouncing back unscathed, but struggling well, effectively working through and learning from adversity, and integrating the experience into life's journey (Walsh, 2012).

A family resilience orientation involves a crucial shift in emphasis from family deficits to family challenges, with conviction in the potential inherent in family systems for recovery and positive growth out of adversity. By targeting interventions to strengthen key processes for resilience, families become more resourceful in dealing with crises, navigating disruptive transitions, weathering persistent stresses, and meeting future challenges. Although some families are more vulnerable or face more hardships than others, all are seen to have potential for gaining resilience in meeting their challenges, forging varied pathways. Beyond coping, adaptation, or competence in managing difficulties, resilience processes enable transformation and positive growth, which can emerge out of experiences of adversity. This research-informed conceptual framework can usefully be integrated with many strengths-based practice approaches and applied with a wide range of adverse conditions, with attunement to family and cultural diversity. Resilience-oriented services foster family empowerment as they bring forth shared hope, develop new and renewed competencies, and strengthen family bonds (Southwick et al., 2011).

2.6 Keys to resilience

There are particular keys to resilience. These are identified by Walsh (2006) in three different areas which are family belief systems, family organization and resources, and family communication.

2.6.1 Family belief systems

Family resilience is fostered by shared beliefs that help members make meaning of crisis situations; facilitate a positive, hopeful outlook; and provide transcendent or spiritual values and purpose. Families can be helped to gain a sense of coherence (Antonovsky & Sourani, 1988) by recasting a crisis as a shared challenge that is comprehensible, manageable, and meaningful to tackle. Normalizing and contextualizing members' distress as natural or understandable in their crisis situation can soften their reactions and reduce blame, shame, and

guilt. Drawing out and affirming family strengths in the midst of difficulties helps to counter a sense of helplessness, failure, and despair as it reinforces pride, confidence, and a "can-do" spirit. The encouragement of family members bolsters efforts to take initiative and persevere in efforts to overcome barriers. As such, therapists also can help family members focus efforts on mastering the possible and accept that which is beyond their control. Spiritual or religious resources, through faith practices such as mediation or prayer and religious or congregational affiliation, now have empirical support for their healing power. Many find strength and recovery through more soulful connection with nature or through artistic expression. Although spiritual resources have been largely neglected in clinical practice, they can be tapped as wellsprings for resilience (Walsh, 1999).

2.6.2 Family organization

In family organization, resilience can be fostered through flexible structure, shared leadership, mutual support, and teamwork in facing life challenges. Families in transition are assisted in navigating disruptive changes and structural reorganization, as with the loss of a parent or with post-divorce and stepfamily reconfigurations. Clinicians can help families counterbalance disorienting changes with stability. Especially valuable are strategies that reassure children and other vulnerable family members by coaching behaviours that reflect strong leadership, security, continuity, and dependability.

2.6.3 Communication

Communication processes that clarify ambiguous situations, encourage open emotional expression and empathetic response, and foster collaborative problem solving are especially important in facilitating resilience. Therapeutic efforts are future directed, helping families "bounce forward" (Walsh, 1998b). Families become more resourceful when interventions shift

from a crisis-reactive mode to a proactive stance, anticipating and preparing for the future. Most important, interventions help families in problem-saturated situations to envision a better future and take concrete steps toward achieving their hopes and dreams (Walsh, 1998b).

2.7 Prominent family resilience factors

A review of recent research and literature recognizes additional recurrent and prominent attributes among resilient, healthy families (Black & Lobo, 2008). These factors include: a positive outlook, family member accord, financial management, time together, mutual recreational interests, routines and rituals, and social support.

2.7.1 Positive outlook

An optimistic confidence in overcoming odds lies at the heart of resilience. Rutter (1987) identified an approaching style to new situations and a sense of humour as important features in viewing crises as meaningful or comprehensible. The importance of optimism was emphasized again by Rutter in 1993:

Almost certainly, it is not that there is one "right" way of thinking about things, or one optimal style of coping. Rather, what seems important is to approach life's challenges with a positive frame of mind, a confidence that one can deal with the situation, and a repertoire of approaches. (p. 630) Bradbury and Karney (2004) reported that displays of positive emotions (such as affection, humour, offering positive solutions, and accepting suggestions) were more influential in marital satisfaction than problem solving or communication. Hope for the future was found to foster resilience, measured by the avoidance of risky behaviours, among 443 impoverished inner-city black female adolescents and to thwart risk-taking behaviours in Aronowitz and Morrison-Beedy's (2004) study. Affirming strengths and possibilities, maintaining courage and hope, and remaining optimistic have been shown to be effective

methods of coping for families (McCreary & Dancy, 2004). Maintaining a positive outlook does not mean that negative behaviour is ignored, but rather that various viewpoints and each member's opinions are examined to identify underlying problems, always augmenting positive aspects (Leon, 2003). For example, moving to a new area could be seen as a burden for some families and met with resistance. For families with a positive outlook, the relocation could be seen as a welcome challenge and an opportunity to interact with new people and locations.

2.7.2 Family member accord

The premise of the systems approach to families is that the whole and its parts must be connected and content for both to continue. A family system works when its members feel good about the family, needs are being met, and the development of relationships flows smoothly (Olson, 2000). Family cohesion enhances family confidence that problems are comprehensible, manageable, meaningful, and that higher levels of reorganization and adjustment are reached after the crisis (McCubbin et al., 2002; Walsh, 1998). Well-functioning family members tend to interact on a daily basis with nurturance and compliments, and they reinforce each-others' efforts (Walsh, 2003a). The comfort and security of a resilient home may be seen as a protective and secure base for all family members (Bowlby, 1982). In times of crisis, the family's pulling together and turning to each other for support has been found to be one of the most important recovery factors in resilience (White et al., 2004). Children seem to do better when there is a fit between the temperament, personality, and needs of the children and the style of parenting they receive. This fit is best achieved when parents assume loving and leadership roles (Pruett et al., 2003). Discipline in resilient families tends to be authoritative and predictable, with mutual respect for all family members. An authoritative parenting style is characterized as warm and affectionate, clear and firm about expectations, yet not too rigid. Consistently practiced, this type of parenting and discipline has been found to achieve positive child outcomes, such as perseverance, better school outcomes, and the development of resilience into adulthood (Masten, 2001). Conversely, too-strong parental control, or an authoritarian style, has been associated with poor child outcomes, such as behavioural problems, poor academic achievement, and less initiative (Morris et al., 2002).

2.7.3 Financial management

Financial pressure in families contributes to family tension and stress and has been shown to have a pervasive effect on emotional well-being and interpersonal relationships (Walsh, 2002). Family research suggests that economic pressures will first affect the emotional lives and marital interactions of adults and then diffuse into the caretaking environment of the children (Mederer, 1999). Poor families potentially confront multiple stresses, such as unemployment, substandard housing, lack of health care, crime, violence, and substance abuse (Staveteig & Wigton, 2005). The combination of psychological, social, and economic burdens of poverty renders families more at risk for multiple problems and crises due to forces beyond their control (Walsh, 1998). Despite these pressures, many low-income families not only meet basic needs, but are able to avoid violence and crime involvement, keep their children in school, engage their children in enriching activities, and maintain family cohesion (Woolley & Grogan-Kaylor, 2006). "Sound decision-making skills for money management and satisfaction with economic status can contribute to family well-being" (McCubbin & McCubbin, 1988, p. 248).

2.7.4 Family time

Spending time together during family meals, chores, and errands plays an important role in creating continuity and stability in family life (McCubbin & McCubbin, 1988). Meal time appears to be the main activity families share together, totalling about 2 hours a week, followed by television viewing (Crouter et al., 2005). Time spent as a family unit has been shown to

decrease risky behaviour significantly in teenaged children (Milkie et al., 2004). Unfortunately, family time appears to be dwindling as a result of increased parental demands and responsibilities. More than half of American families feel they spend insufficient time with their children, according to Milkie and colleagues. Full-time employed mothers were reported to spend, on average, 50 hours a week with their preschool children, with fathers spending 33 hours a week. In contrast, non-employed mothers spend 63 hours a week with children, with time declining as the child enters adolescence and the young adult years. A common parenting phenomenon is time strain, conceptualized as the lack of spontaneity to respond to children's needs, fatigue, and inability to disconnect feelings from work (Milkie et al., 2004). Both employed mothers and fathers feel time strain, particularly when children younger than 6 years old are in the household (Voydanoff, 2004). Employer-sponsored efforts to reduce work-related family strain, such as child care and family-supportive programs, have not been shown to increase quality family time for working parents (Voydanoff, 2004).

Time strain will likely be a growing family concern with the trend toward dual earners and single-parent households. An effective way to reduce time strain and increase family time is to include the family in constructive family activities including housekeeping routines. Children may be assigned age-appropriate meaningful tasks, such as caring for the family pet, where they can learn skills, abilities, and empathy. Commute time spent together in the car or bus can provide opportunities for sharing family time through conversation, making future plans, or participating in fun learning activities.

2.7.5 Shared recreation

The multiple family benefits of shared recreation and leisure time have been found to facilitate family health. Enjoyable family time can yield attachments, intrinsic rewards, happiness, learning, humour, and the pleasure of shared experiences (Russell, 1996). Mactavish and

Schleien (1998) examined parents' perceived benefits of family recreation and concluded that "shared recreation was especially helpful in developing social skills, such as learning to problem-solve, to compromise, and to negotiate" (p. 221). A 5-year longitudinal study of 280 randomly selected couples conducted by Hill (1988) suggested that families who play together tend to stay together. There was a significant relationship between shared leisure time and lower divorce and separation rates. Interestingly, the presence of children in the household significantly reduced the amount of leisure time that is viewed important in counterbalancing stress and strains. Zabriskie and McCormick (2001) identified two types of family recreation: (a) core, or routine and common activities, such as playing a board game after dinner; and (b) balance, or novel activities, such as bowling. Both types of recreation were found to predict family adaptability and cohesion. Lareau (2002) argued that middle-class families tend to make deliberate efforts to spend time together. Working-class parents, however, felt it is up to the children to devise their own recreational pursuits; yet these parents strive to maintain safe play environments. Both approaches may be viewed as family strengths that promote either fun family time or child autonomy. In recent years, screen media have taken precedence as a leisure activity for many families. American families appear to be spending more time with television, video games, and the computer (Bagley et al., 2006). Although some educational and recreational benefits with moderate screen media usage have been found, the difficulty arises when family members sacrifice interaction time and physical exercise in favour of passive viewing. Kubey and Csikszentmihalyi (2002) described an "addiction" to excessive television viewing that can disrupt family cohesion. Moderate screen media use can be fun and educational for the family, but not at the sacrifice of true interactional activities.

2.7.6 Routines and Rituals

Based on reviews of 50 years of research on multicultural family routines and rituals, Fiese and colleagues (2002) reported common strengths that these embedded activities contribute to close family relationships, child socialization, and mitigation of stressful situations. Routines are defined as momentary time commitment tasks requiring little conscious thought. Rituals, however, involve symbolic communication with enduring, affective, and generational transmission. Research has indicated that families who practice routines and rituals are associated with better child outcomes and family stability than families lacking routines and rituals. Routines are often dismissed in the event of a family crisis, such as marital separation or illness (Walsh, 1998). The loss of a comforting practice, such as a habitual bedtime routine or family dinner times, can be unsettling to children. Well-functioning families attempt to maintain family routines in the midst of turbulence to provide the family with a feeling of predictability, cohesion, and comfort (Walsh, 1998). Young and older generations alike find rituals and celebrations to be a way to maintain family-of-origin values and to build and maintain successful families through the generations (Fiese et al., 2002).

2.7.7 Support network

A family's community and social support system can offer a rich, protective sense of belonging and cohesion. There is a growing appreciation for the broader view of resiliency as a complex interaction with community networks (Luthar et al., 2000). Resilient families not only attain social support from their communities, but also give back to the community (Patterson, 2002). Extended kin and social networks can interact reciprocally with families to provide information, services, respite, avenues to contribute to the welfare of others, and companionship (Seccombe, 2002). Children who perceive their community and neighbourhood as safe and satisfying have been found to benefit through improved academic

grades and fewer problem behaviours than those who see their neighbourhood as threatening (Woolley & Grogan-Kaylor, 2006). Particularly for families living in impoverished or isolated conditions, the availability and use of quality support systems can dramatically increase positive outcomes, such as perseverance, hope, education, and companionship (Orthner et al., 2004). If parents are not able to provide emotional support for children, other kin or mentors may step in to offer the children support, positive role models, and protection (Aronowitz & Morrison-Beedy, 2004). Healthy families have the strength to admit when they need help and are more likely to turn to extended family, friends, neighbours, community services, or counselling (Walsh, 1998). Unfortunately, there appear to be two trends diminishing a family's interaction with the community. One is the tendency to withdraw from society; the other is a lack of available services. The trends toward relocation and smaller family size in modern society have frayed kin and community connections, requiring the establishment of new forms of social support (Hanson, 2001a). Americans are more socially isolated than they were just 20 years ago, according to McPherson, Smith-Lovin, and Brashears (2006), who recently replicated a 1985 social survey. During the past two decades, adults reported a 50% reduction in intimate social ties, both within and outside the family, and 25% of the adult Americans surveyed reported that they do not have a confidant. The researchers contemplated that increased professional responsibilities, television viewing, and technology dependence may be some of the recent socio-demographic trends to blame for growing social isolation. Additionally, a recent tendency for families to become privatized, with reduced concern and responsibility for their neighbours, has been described. Toth, Brown, and Xu (2002) expressed this dichotomy as a public-private division; families feel a time pull for either private or public life, often with a perceived inability to focus on both. When a balance was made between public and private realms, community satisfaction was found to be positively correlated with family life satisfaction. It may be, however, that families are not becoming privatized, but

rather the cause of social isolation lies in the paucity of community services. Walsh (1998) cautions:

One must beware of the myth of family self-reliance that has grown out of society's individualistic strain. The major problems of families today largely reflect difficulties in adaptation to the social and economic upheavals of recent decades and the unresponsiveness of larger community and societal institutions (p. 101). All families benefit from community resources, but those with a host of other social problems can particularly benefit from external support (Staveteig & Wigton, 2005). The poor have all the personal and family problems everyone else has, along with complications of poverty, isolation, and a loss of community. Family isolation with a lack of social support can erode resilience, particularly under stressful conditions (Luthar, 1999).

When these family resilience factors are applied and family members are aware of their positive effects, they are able to contribute positively to learner academic performance. Academic success is critical to developing a healthy and positive motivational orientation toward learning in school. It is therefore worthwhile to ensure that that these factors are intertwined with learners' well-being.

2.8 Learner academic performance

Academic achievement is assessed through measuring the degree of each student's institutional learning through his/her scores (Seif, 2001). DiPerna and Elliot (2000) state that the multidimensional characteristics of a learner – including skills, attitudes, and behaviours - factor into their academic success. These characteristics can be separated and considered in one of two primary domains: academic skills or academic enablers. Academic skills are both the basic and complex skills (e.g., reading, writing, calculating, and critical thinking) needed to access and interact with content-specific knowledge. Academic enablers, however, are the

attitudes and behaviours e.g. interpersonal skills, motivation, study skills, and engagement, which a learner needs in order to take advantage of education.

Individual resources associated with academic success include cognitive abilities, motivation, and beliefs. IQ is one of the most powerful predictors of academic success. Yet academic competence also is influenced by beliefs and attitudes about school (Stevenson et al., 1993), self-perceptions about one's academic abilities (Greene & Miller, 1996), and motivations to succeed (Henderson & Dweck, 1990). Successful students typically attribute their successes to hard work and their failings to lack of effort. Students who maintain a belief that ability and performance are fixed tend to have lower achievement (Stipek & Gralinski, 1996) in comparison to children who believe that performance is a function of effort and hard work (Stevenson et al., 1993).

Despite widespread acceptance of the notion that improving student performance may have a high economic and social payoff, policy analysts in all countries have surprisingly limited hard data on which to base educational strategies for raising achievement. In South Africa this question is all the more pressing. South African students score at low levels in mathematics and language tests even when compared with students in other African countries (van der Berg and Louw, 2006). While some reasons for this poor performance may be evident, and there is widespread agreement that the main challenge in South Africa is the quality of education, there is little empirical analysis that helps policy makers understand the low level of student performance in South African schools or how to improve it.

As a first step toward an empirical approach to unpacking the factors contributing to low levels of learning in South African schools, the Human Sciences Research Council in partnership with a consortium of South African universities and researchers at the School of Education at Stanford University engaged in a small scale empirical pilot study that focuses on the role that

teacher skills and practice play in South African students' learning within the socioeconomic and administrative conditions in those schools (and South African society more broadly). In further studies more energy was focused in measuring opportunity to learn. Opportunity to learn is undoubtedly also an important factor in explaining student learning differences (Carnoy et al., 2008).

2.8.1 Course grades

Although performance on standardized tests receives the greatest attention in discussions of students' academic performance, teachers' evaluations of performance as indicated in course grades represent a common metric of student performance that often is more directly tied to the day-to-day business of teaching and learning than are annual standardized test scores. Grades serve a number of important functions. They communicate to students and parents information about students' mastery of course content. In high school, a passing grade also is the criterion for a course's contributing to accumulated credit for graduation. Finally, grades provide information for consideration in college admissions (Polloway et al., 1994).

However, as a measure of academic performance, teacher-given grades have well-known limitations. Grades are composite measures that account not only for students' content mastery but often for other factors, such as their class participation, attitudes, progress over time, and attendance. Both general and special educators are known to consider these various factors when grading, but to emphasize different factors. For example, special education teachers are less likely than general educators to consider homework or attendance to be important in grading student performance, but are more likely to consider in-class participation to be important (Blackorby et al., 2003).

2.9 Family resilience and children's learning

From the family resilience perspective, the family-school relationship becomes a collaborative one in which the educator recognizes that successful interventions to enhance children's learning depend more on tapping into a family's resources than on specific change techniques. As a result, assessment and intervention efforts are redirected from looking at how children's learning problems are caused to looking for family strengths, or resiliencies, that can be employed to resolve a child's problem. From this positive, future-oriented stance, educators and family members work together to find new possibilities for growth and to overcome impasses to children's learning and development.

A family resilience perspective considers each interaction between home and school as an opportunity to strengthen a family's capacity to overcome adversity and successfully rear its children (Walsh, 1998, 2003). Two basic premises guide this resilience theory approach. The first premise is that while stressful crises and persistent economic, physical, and social challenges influence the whole family and its capacity to successfully rear its children, key family processes mediate the impact of these crises and the development of resilience in individual member and in the family unit as a whole. A second premise is that while family processes mediate how children are prepared to participate in school, these key family processes can be strengthened by the way the school responds to families. As the family becomes more resourceful, its ability to rear its children is enhanced. As a result, each family-school intervention also can be a preventive measure (Amatea et al., 2006).

McCubbin and McCubbin (1996) developed a Resiliency Model of Family Stress, Adjustment and Adaptation, in which they depicted the protective role that particular family characteristics play in facilitating a family's recovery from stressful life experiences. In contrast, Conger and Conger conceptualized family resilience in terms of specific processes – such as the

development of close and supportive family ties – that evolve over time in response to a family's specific context and stage of development. Similarly, Seccombe (2002) identified resilient families as having clear-cut expectations for their children, creating routines and celebrations, and sharing core values.

2.10 Conclusion

It can therefore be deduced that family resilience is crucial to the academic performance of learners in the phase of middle childhood. The family operates as a system and is viewed as a structure that is constantly changing. The purpose and value of family is widely spread and touches most of human existence. This chapter aimed to examine literature regarding family resilience, viewed in the context of family resilience theory, which was the conceptual framework of this study. Academic performance and how it is enhanced through family resilience, was examined. The following chapter provides an overview of the literature, based on the variables that this research examined.

WESTERN CAPE

47

CHAPTER 3

LITERATURE REVIEW

3.1 Introduction

This chapter provides a critical insight into some of the research conducted, as well as an analytical appraisal of available literature, both internationally and locally. This review of the literature is presented in five areas. The first section of the chapter looks at middle childhood and its cognitive, social and emotional functioning. This is followed by discussions on assumptions of family life, family structure and family resilience. The final section looks at academic performance. All these concepts are discussed as an in-depth overview. This chapter then ends with a conclusion.

3.2 Middle childhood

Middle childhood is one of the main stages of human development, marked by the eruption of the first permanent molars around 6 years and androgen secretion by the adrenal glands at about 6–8 years (Bogin, 1997). In middle childhood, body growth slows considerably, usually following a small mid-growth spurt. At the same time, muscularity increases and the body starts accumulating fat (the *adiposity rebound*; Hochberg, 2008), while sex differences in body composition become more pronounced (Del Giudice et al., 2009; Wells, 2007).

In biological terms, middle childhood corresponds to human *juvenility* - a stage in which the individual is still sexually immature but no longer depends on parents for survival. Social learning in juvenility can be understood as investment in *embodied capital* - skills and knowledge that cost time and effort to acquire, but increase an individual's performance and reproductive success (Kaplan, Hill, Lancaster, & Hurtado, 2000).

3.2.1 Cognitive functioning

Social learning is universally recognized as a key evolved function of middle childhood and is enabled by a global reorganization of cognitive functioning known as the five-to-seven shift (Weisner, 1996). By age 6, the brain has almost reached its maximum size and receives a decreasing share of the body's glucose after the consumption peak of early childhood (Giedd & Rapoport, 2010; Kuzawa et al., in press). However, brain development proceeds at a sustained pace, with intensive synaptogenesis in cortical areas (grey matter) and rapid maturation of axonal connections (white matter; Lebel, Walker, Leemans, Phillips, & Beaulieu, 2008).

3.2.2 Social functioning

Researchers across multiple disciplines share an interest in examining how social and contextual assets are related to developmental outcomes, such as physical and mental health, social and emotional competencies, and subjective well-being (Oberle et al., 2011). In this regard, it has been shown that social relationships are strongly related to children's developmental health and well-being across virtually all domains of functioning - social, emotional, physical, mental, and cognitive (Luthar 2006). In addition, children during middle childhood have a need to develop competency, particularly in domains that are societally valued, such as social skills, academic achievement, music, or sports. Accordingly, children who have opportunities to develop such competencies in their family, school, and community environments show higher levels of well-being than children who do not have such opportunities (Masten & Coatsworth 1998).

Middle childhood is also a phase of fast physical growth, and children's physical health and well-being during that period is related to their nutrition (Connell et al., 2005), their sleep

quality and pattern (Smaldone et al., 2007), and their physical activities and exercise (Norris et al., 1992). Furthermore, research suggest that indicators of quality time with one's family (e.g., regular family meals; Eisenberg et al., 2004) and certain after-school activities (e.g., sports, music) are positively related to children's well-being (Donaldson & Ronan 2006). In sum, research has identified a number of social and contextual assets that are significantly related to children's overall health and well-being (cf. Theokas et al., 2005). There are numerous initiatives that aim to translate the cumulative research evidence on children's well-being and social and contextual assets into effective community practices, programs, and policies (Durlak et al., 2011). Also, a number of research initiatives attempt to develop indicator systems of children's or adolescents' well-being in order to raise awareness and visibility of the topic in public and political debate, and to allow for regional or cultural comparisons of child and adolescent well-being (Casas et al., 2011).

Early childhood behaviour problems have been shown to have detrimental and enduring effects UNIVERSITY of the in middle childhood and adolescence (Shaw & Gross, 2008). Several dimensions of parenting have been found to play a central role in the development of early childhood disruptive problem behaviours (Shaw et al., 2003), including both hostile and over-controlling practices (Shaw et al., 2003), and infrequent positive and proactive strategies (Gardner et al., 2007). An example of reciprocal influences is described by Shaw and Bell (1993), in which emotionally reactive young children evoke harsh parenting practices, thereby stifling children's acquisition of emotion regulation strategies, and further affecting parents' behaviour toward their children. This dynamic in families is postulated to leave young children with impaired social skills and failed individuation from parents, leading to difficulties with peer relationships (Scaramella & Leve, 2004) and acrimonious teacher–student relationships. Coercion theory has informed several behavioural parenting interventions that are effective at reducing growth of child

behaviour problems (Dishion et al., 2008). Parenting interventions initiated during the toddler period, when oppositional and aggressive behaviours increase as children become more mobile yet lack sufficient concomitant development in cognitive understanding of basic principles (e.g., gravity, electricity), specifically focus on increasing positive behaviour support to reduce the frequency of coercive parent—child interactions (Dishion et al., 2008). Given the presumed bidirectional nature of influence in the coercive cycle, early childhood interventions that address and attenuate the harsh parenting styles often elicited by negative emotional reactivity (Shaw et al., 2000) should also promote parent well-being. Developmental research attests to the impact that infant and early childhood behaviour problems have on parenting effectiveness, parent well-being, and marital functioning (Crouter & Booth, 2003).

3.2.3 Emotional functioning

The transition to middle childhood is marked by a simultaneous increase in perceptual abilities (including a transition from local to global visual processing), motor control (including the emergence of adult-like walking), and complex reasoning skills (Bjorklund, 2011; Poirel et al., 2011; Weisner, 1996). The most dramatic changes probably occur in the domain of self-regulation and executive functions: Children become much more capable of inhibiting unwanted behaviour, maintaining sustained attention, making and following plans, and so forth (Best, Miller, & Jones, 2009; Weisner, 1996). Parallel improvements take place in mentalizing (the ability to understand and represent mental states) and moral reasoning, as children become able to consider multiple perspectives and conflicting goals (Jambon & Smetana, 2014; Lagattuta, Sayfan, & Blattman, 2009).

3.3 Assumptions of family life

Across cultures and history, families are the most enduring institutions in the world. Strong families are identified by their ability to maintain a family equilibrium, drawing upon individual and joint strengths to cope effectively. The concept of family equilibrium does not imply a static pattern of family functioning. Rather, strong families are dynamic and responsive to changing needs, developmental tasks, and challenges. Strong families celebrate their successes and learn from their failures. These families have clearly defined roles, especially with regard to the parent-child relationship (Guilfoyle, Goebel, & Pai, 2011).

Families are also complex, and must be understood within numerous interdependent layers, including each individual family member, dyadic relationships, broader family functioning, and the culture in which they live (Rasbash, Jenkins, O'Connor, Tackett, & Reiss, 2011). This evolution to family resilience focuses on understanding what the family does well, the positive ways the family functions, and attempts to build on the positives to help the family improve not only its overall functioning, but also its ability for problem-solving, coping, and adjusting (Frain et al., 2008). Family resilience is grounded in family systems theory, which focuses on the entire family network and the way in which the family functions and adapts to adversity or stress (Walsh, 2003a). Walsh (2003a) noted that families have built in processes that enable the family members to handle stress, come together during a crisis, and move towards optimal adaptation.

McCubbin et al., (1996) stated that there are five assumptions of family life: (a) hardships and challenges are a part of the family life cycle; (b) families have patterns and ways of functioning to protect the family during transition and change, and to foster growth and development in all family members; (c) families have patterns of functioning to provide protection from major stressors and change; (d) families draw from and contribute to the community around them;

and (e) families work to restore order, harmony, and balance even in the midst of crisis or change. As families advance throughout the family life cycle, they are faced with multiple changes and challenges. Some of these changes or challenges can be positive such as the birth of a child or the moving out of an adult child. However, there are changes and difficult challenges that families experience throughout the family life cycle. These changes include the death of a family member (whether parent or child) or the loss of a job. The changes within a family that cause disruption and stress are known as stressors (McCubbin et al., 1996).

3.4 Family structure and well-being

Social scientists have long been concerned with the determinants of adverse social phenomena, such as interpersonal conflict and depression on individuals that have little apparent capacity to shape these outcomes. In the 1970s, the research began to shift its focus to understanding why some persons succeed in spite of adverse circumstances. This change in focus gave rise to studies on what is now referred to as 'resilience', which until recently has predominantly focused on characteristics of the individual. For instance, a review of the evidence shows that since the 1970s many children who were raised in deprived, adverse environments actually went on to become successful and loving individuals (Luthar et al., 2000). Research into resilience has since expanded to include various age groups and disadvantaged states, such as chronic illness (McCubbin et al., 2002), mental health issues (Jonker & Greeff, 2009), poverty and violence (Luthar, 1999) and, catastrophic life events (Clay, 2002). Included in this array is family resilience (FR) (Patterson, 2002a).

Research on the impact of family structure on child wellbeing came to the forefront of social science soon after the spike in divorce rates in the 1970s, and researchers worked to identify how children from divorced and step-families fared compared to children from intact families. As the rates of non-marital fertility have drastically increased over the past 20 years, more

recent research efforts have focused on understanding the life chances for children born to single mothers and cohabiting couples.

Older children and adolescents have greater cognitive capabilities to understand the impacts of family structure change, and they also have more extra-familial resources to rely on for support. The adjustment of younger children, however, depends heavily on their experiences within the family and the resources available to them through the family (Hetherington, 1992). For example, younger children are more likely than adolescents to have a stronger emotional bond with a new stepparent and are more likely to consider the stepparent the "real" parent (Cherlin & Furstenberg, 1994). In addition, children who have been in non-traditional families for longer periods of time have had a longer opportunity to adjust to the new family and become settled in their role in the family system (Hetherington, 1989). Thus the adjustment of both the parents and the children in non-traditional families may differ based on the developmental stage of the child during the transition to stepfamily life.

While there are clear research efforts to examine the impacts of certain family structures on child well-being, many children experience multiple types of family structure throughout their childhood. Understanding the challenges children face in each type of non-traditional family is necessary, but it is important to remember that these family types are not always experienced in isolation, and their co-occurrence has the potential to produce further challenges for children. The aggregate experience of multiple changes in family structure can produce unique challenges for children who have high levels of family instability.

The family system has perhaps the greatest impact on individual development across all stages, influencing not only individual and family life, but also that of the community in which it is found. Despite being confronted by more extreme and disruptive stresses and demands than ever before, the family system is still expected to be adaptive, competent and resilient, and to

provide a protective, growth-enhancing environment for its members (Frude, 1991). It is evident that problems and challenges, such as gross political atrocities, soaring divorce rates, diverse family forms and changing economies, do not necessarily evoke psychopathological responses. Instead, as Jenkins (1997) stated in his writing on resilience among Bosnian refugees: "what we have seen... are truly remarkable instances of human resilience and irresistibility" (p. 40).

3.4.1 Family processes

The family is one of the most important contexts for child development (Bronfenbrenner, 1986). Family systems theory posits that family members are part of an interdependent system in which each individual impacts upon and is impacted by every other individual in the family (Minuchin, 1974). In addition, the family is comprised of subsystems between the individual members: the executive subsystem (the husband and wife), parent-child subsystems, and sibling subsystems. Families typically maintain a high degree of orderliness in the way that they interact with each other, but a disturbance or change in one family member or one subsystem impacts all other members and subsystems. In this way, a change in the executive subsystem (e.g. divorce or remarriage) impacts parent-child relationships and also sibling relationships (Minuchin, 1974). Family transitions and changes upset the homeostatic functioning of the family system, forcing families to renegotiate communication rules and boundaries. Though the renegotiation phase can be tumultuous and lengthy, adaptation to the new system at even just one level can influence adaptation among all the subsystems (Hetherington & Jodl, 1994). It should be noted that the members of the family are not limited to residential members; non-residential parents (usually fathers) play a vital role in the maintenance of the family system through interactions with the residential parent and child visitation.

The systems perspective supports the notion that the structure of the family influences child wellbeing through its impact on overall family functioning. Though few studies cite family systems theory, there is a growing body of literature that examines the mediating role of family processes on family structure and child wellbeing. Most studies investigate various family processes independent of each other; family processes are typically operationalized as dyadic measures of family functioning, including parent-child relationships, the adults' relationship quality with each other, and co-parenting.

The construction and maintenance of high quality parent-child relationships is vital for child wellbeing. Positive communication, praise, parental warmth, and better monitoring are related to healthy behaviours, academic engagement, and positive emotional adjustment (Moore, Chalk, Scarpa, & Vandivere, 2002). However, non-traditional families often have to struggle to maintain high quality parent-child relationships when one of the parents is non-residential. There is a great deal of variability in the relationship quality between non-residential fathers and their children (King, 2006), and the closeness of the relationship appears to be more dependent on the quality of shared time rather than on the frequency of visitation (King, 2006). Non-residential fathers often take on the role of their child's friend and playmate (Hetherington & Jodl, 1994), though children have better adjustment when the non-residential father maintains his authoritative parenting role (Hetherington, Bridges, & Insabella, 1998). Nonresidential father involvement is also dependent on whether or not the father re-partners. For instance, there is evidence to suggest that child support payments decrease once the father has a new biological child to support (Manning, Stewart & Smock, 2003), and that having a new romantic partner can decrease the level of positive engagement between the father and nonresidential child (Gibson-Davis, 2008).

The relationship between the adults in the household also has the potential to impact child wellbeing. The relationship quality of a married couple even before a child is born is related to the child's attachment to the mother and dependency on the father (Howes & Markman, 1989). Children in families with high levels of marital conflict have more psychological problems than those from low-conflict (or even divorced) families (Booth & Amato, 2001). When marital quality (or relationship quality among unmarried partners) is low, parents are more likely to be depressed, anxious, or angry, and their mental health may interfere with their parenting abilities (Hetherington, 1989).

In addition to the relationship quality of the household members, co-parenting – agreements about how to parent – may be particularly important for non-traditional families in which the parents live in separate households. Indeed, co-parenting has been found to be a strong predictor of both parent and child wellbeing (Feinberg, Kan, & Hetherington, 2007). However, the co-parenting efforts of the mother and non-residential father are not well understood. In a study with divorced parents (not yet remarried), Maccoby, Dpener, & Mnookin (1990) found that the level of co-parenting between the mother and biological father is largely dependent on the father's involvement with the child, as well as the level of inter-parental hostility between the mother and non-residential father. As previously mentioned, non-residential fathers often assume the role of a friend to their child rather than a disciplinarian (Hetherington & Jodl, 1994), perhaps removing themselves from the co-parenting process entirely. This separation from the parenting practices of the residential parent can lead to parallel parenting rather than co-parenting, resulting in a lack of communication regarding common parenting strategies (Hetherington, 1998).

Although each of these processes can be individually linked to child wellbeing, they cannot be considered in isolation. Following a family systems perspective, all dyadic relationships are

influenced by other each other, and the functioning of the family as a whole is determined by the combination of the individual relationships (Minuchin, 1974). Research on stepfamilies provides an example of evidence for this perspective: following remarriage, high marital quality predicts better parent-child relationship quality (Hetherington & Jodl, 1994). While the identification of which dyadic relationships have the potential to impact child wellbeing contributes to our understanding of family processes, researchers must recognize that the overall functioning of the family is dependent on all of the relationships within the family and how they interact to impact the family as a whole.

3.4.2 Positive and involved parenting

Parental involvement is the extent to which parents participate in various aspects of their children's lives, and distinguishes between negative and positive involvement. As such, researchers often refer to positive parenting and parental involvement collectively in the examination of children's aggression (Finley, Mira, & Schwartz, 2008).

WESTERN CAPE

Positive parenting techniques are characterised by warm, nurturing, and supportive behaviour towards children, and espouses the tenets of effective discipline, the rights of children and healthy child development (Durrant, 2007). Previous studies have found that an absence of parental warmth in addition to rigid discipline, a lack of positive parental involvement and insecure attachment increase the prognosis for later conduct problems in children (Sanders, 1999). Attachment theory emphasizes the importance of parental emotional sensitivity to effectively interpret and respond to the needs of the child. The more stable the bond between parent and child, the more likely it is that a child will regulate conduct towards others in an empathetic manner (Bowlby, 1969). Based on the positive results of previous interventions, researchers propose that positive parental involvement is very likely to serve as a protective factor against children developing aggressive behaviour (Trentacosta et al., 2008).

3.4.3 Effective discipline

In the public domain, discipline is often construed as being synonymous with punishment, power assertion, and control. However, effective discipline is a positive parenting technique aimed at shaping acceptable behaviour in children and promoting positive engagement with others (Stein & Perrin, 1998). Through effective discipline, children cultivate empathy towards others and learn self-discipline. In addition, before acting on an urge or feeling, children learn to consider the repercussions of a future action (for themselves and others) and are thus more likely to conduct themselves in a socially acceptable manner. The ability to weigh consequences, and self-regulate behaviour supports healthy relationships and social acceptance, and thereby decreases the risk of children developing conduct problems. Furthermore, effective discipline builds a child's self-respect, teaches children to respect others, and assists in building a mutually respectful relationship between parent and child (Durrant, 2007).

3.4.4 Healthy relationships

Through a mechanism psychologists call 'attachment', (Ogilvie et al., 2014) parents set the tone for their children's later relationships and begin to help them to manage their emotions and behaviour (a capacity psychologists call 'self-regulation') (Henry et al., 1996). Both of these are crucial for education (and hence for employment and income), and also for helping children to curb aggressive impulses that could lead to violence (Espelage, 2014).

WESTERN CAPE

Most children are aggressive and demanding at age two, but over time most learn to ask politely for what they need, to take turns and to manage emotions such as anger (Tremblay et al., 1995). Children who do not learn these skills are likely to go on to develop behavioural problems.

Learning to regulate emotions starts with parenting (Sanders et al., 2013). Parents who help toddlers to name their emotions are teaching them to recognise their internal states, which is the first step towards managing emotions. Parents who establish consistent household routines and who use non-violent approaches to discipline, such as praising behaviour that is appropriate and ignoring behaviour that is not, help children to manage their own behaviour and to find appropriate ways to get what they need and want. This is very difficult for parents who are depressed, anxious or struggling to survive.

3.5 Family resilience studies

With the evolving concept of family resilience, the focus is shifting away from identifying individual personality factors toward the crucial influence of positive relationships with family, kin, and mentors. Family resilience examines the product of family relationships (Patterson, 2002). Luthar et al (2000) argued that the dynamic process of resilience is best understood through the context of a broader, inter-relational framework. A family resilience perspective recognizes parental strengths, family dynamics, inter-relationships, and the social milieu. This strength-based approach considers family stresses and challenges not as damaging but rather as opportunities for fostering healing and growth (McCubbin & McCubbin, 1988, 1996; Walsh, 2003b). All families encounter problems, or stressors. "A stressor is a demand placed on the family that produces, or has the potential to produce, changes in the family system" (McCubbin & McCubbin, 1993, p. 28). The problem, demand, or loss has the possibility to either weaken family functioning and relationships or, conversely, strengthen the family as the stressor is subdued with unified action and commitment. Resilient families are strengthened through the use of resilience factors (protective and recovery) in these circumstances. McCubbin and McCubbin (1993) identified both protective and recovery factors that work synergistically and interchangeably to respond successfully to crises and challenges. Protective factors facilitate

adjustment, or the ability to maintain integrity and functioning, and to fulfil developmental tasks. When the family is challenged, recovery factors are called upon to promote the ability to adapt, or rebound, in crisis. In addition to the perceived severity of stressors, a "pile-up" of stressors can tax family protective and recovery factors, with the family relying on some more than others during various family cycle situations (McCubbin & McCubbin, 1993). Family resilience entails more than surviving a crisis, but also offers the potential for growth out of adversity. Weathering a crisis together, a family can emerge more loving, stronger, and more resourceful in meeting future challenges (Walsh, 1998; White et al., 2004). The developmental progression of resilience often emerges with changing life circumstances and new stressors (Luthar et al., 2000; McCubbin et al., 2002). "The phenomenon of resilience requires attention to a range of possible psychological outcomes and not just a focus on an unusually positive one or on super-normal functioning" (Rutter, 1999, p. 120). A repertoire of possible coping factors is exchanged in resilient families. There are times when family demands exceed the family's capabilities. When these imbalances exist, some capabilities may supersede others toward regaining equilibrium. Of course, some family processes fail to adapt with poor immediate and long-term outcomes (Patterson, 2002).

Resilience research has taken many different forms, used many means of assessment throughout, and shown many different key concepts to be of importance. Variations in the source of participant information, type of information obtained, and number of assessments used have also been shown to affect the proportion of individuals classified as resilient in studies (Kaufman, Cook, Arny, Jones, & Pittinsky, 1994). Resilience has been assessed through observable behaviour, meeting major societal or cultural expectations, the absence of psychopathology, the presence of academic or social achievements, or both presence and absence of these variables (Masten, 2001).

Despite flaws in early resilience studies, recent studies continue to corroborate the importance of similar factors such as connections to competent and caring adults in the family and community, cognitive and self-regulation skills, positive views of self, and motivation to be effective in the environment (Masten, 2001). Therefore, it is important to review previous research on resilience to search for themes and common factors to establish and guide both conceptual theories and subsequent studies.

3.5.1 Longitudinal study in children's resilience

Potentially the "founding fathers" of resilience research, Werner & Smith (1977), began a longitudinal study in 1955 with 698 babies born that year on the Hawaiian Island of Kauai. This study is unique in that Werner and Smith were able to begin their study with all pregnancies and births in an entire community. Initially this study began by looking at the children's vulnerability or susceptibility to negative developmental outcomes after exposure to serious risk factors such as prenatal stress, poverty, parental psychopathology, and disruption of their family unit (Werner & Smith).

However, as their study progressed over the next 32 years, they began looking at the roots of resiliency in children who successfully coped with risk factors to determine what factors aided in the recovery of troubled children as they moved from childhood, to adolescence, and finally into adulthood. Resilient children are able to elicit predominantly positive responses from their environment and were found to be stress-resistant whereas vulnerable children elicited negative responses from the environment even in the absence of biological stress or financial constraints (Werner & Smith).

"Resiliency in children is the capacity of those of who are exposed to identifiable risk factors to overcome those risks and avoid negative outcomes such as delinquency and behavioural

problems, psychological maladjustment, academic difficulties, and physical complications" (Rak & Patterson, 1996, p. 368). Children are not defenceless against stressful life conditions. There are many factors which can assist to 'buffer' (Rutter, 1985) children against stress, and which assist them in growing up to be well-adjusted and happy adults, who work well, play well, love well and expect well (Werner in Dahlin et al., 1990, p. 228).

3.5.2 Strength within families

Resiliency relates to a family's ability to adapt to change. Some common changes that families face include parenting children from birth to adulthood, caring for an aging family member, or adjusting to parental deployment (Tusaie & Dyer, 2004). Strong families are able to maintain relative equilibrium as they manage both the more common day-to-day challenges with those that are more unique to the typical family's experience (Landau, 2007). Researchers have identified a number of family characteristics that promote resilience to change (Bermudez, 2013). These resilience-promoting characteristics can be seen throughout the family system. Healthy couples are able to be flexible and adapt as a team to new circumstances in the event of transitions or change (Gottman, 2011). Within the parent-child relationship, healthy parents encourage children's adaptive responses to change, are responsive to children's distress, and access needed social and economic resources that enable the family to maintain equilibrium (Mikulincer & Shaver, 2007). Resilience is vital for individuals and families to be able to deal with inevitable changes that will happen. Strong families will strive to be resilient in adapting to everyday changes and not just when extreme situations or traumas occur.

3.6 Academic performance

Academic performance is the multidimensional characteristics of a learner – including skills, attitudes, and behaviours – that factor into their academic success. These characteristics can

be separated and considered in one of two primary domains: academic skills or academic enablers (DiPerna & Elliot, 2000; Elliot & DiPerna, 2002). Academic skills are both the basic and complex skills (e.g., reading, writing, calculating, and critical thinking) needed to access and interact with content-specific knowledge. Academic enablers, however, are the attitudes and behaviours (e.g., interpersonal skills, motivation, study skills, and engagement) that a learner needs in order to take advantage of education.

3.6.1 Predictors of academic performance

There is an international consensus that intelligence is an important predictor of academic performance. Recent research showed that there are also non-cognitive factors, besides the general intelligence, responsible for high academic performance. A critical factor emerging from recent studies that significantly impacts the academic, social and professional adaptation has proven to be the emotional intelligence (e.g. Cherniss, Extein, Goleman, Weissberg, 2006). The results point to the emotional intelligence as a very powerful potential mediator of school outcomes (e.g. Parker et al., 2004; Marquez, Martin & Brackett, 2006; Mestre, Guil, Lopes, Salovey & Gil-Olarte, 2006). The proven benefits of high emotional intelligence in academic context are: coping with academic stress (Petrides, Frederickson & Furnham, 2004), improving overall school climate (Jensen & Freedman, 2006); predicting children's ability to learn and solve problems non-violently (Zins, Weissberg, Wang & Walberg, 2004); decreasing school dropouts (Parker, Hogan, Eastabrook, Oke & Wood, 2006); reducing risky behaviours and increasing prosocial ones (Durlak& Weissberg, 2005 cited in Cherniss et al., 2006; Petrides, Sangareau, Furnham & Frederickson, 2006; Trinidad & Johnson, 2002). The emotional intelligence is considered crucial for the academic and later social success in pre-school childhood (Denham, 2006; Arsenio, Cooperman & Lover, 2000).

3.6.2 Importance of academic performance

Academic performance in the classroom is the end product of many other behaviours. For example, obtaining a good grade after answering examination items is the result of effective performance studying, managing goal conflicts, coordinating work with classmates, seeking additional information, negotiating with peers and faculty, avoiding counterproductive behaviours, handling finances, and structuring effective communications (e.g., Kuncel, Campbell, Hezlett, & Ones, 2001; Reilly, 1976).

During middle childhood, children progressively move from home into wider social contexts that exert important influences on their cognitive, behavioural, and socio-emotional development. In particular, the commencement of formal schooling initiates a series of new life experiences for children. School experiences encourage the development of intellectual and interpersonal competencies, and introduce the child to new social roles wherein status is conferred based upon competence and performance (Higgins & Parsons, 1983).

WESTERN CAPE

Research on children who experience difficulties with learning early in school has shown that these children are at increased risk for behavioural, academic, and psychiatric difficulties contemporaneously and subsequently – such children are also particularly likely to be retained in a grade and to later drop out of school prior to the completion of high school (Alexander, Entwisle & Horsey, 1997). The frequent feelings of frustration and incompetence that accompany a child's lack of success early in her school career may coalesce over time into a negative pattern of adaptation towards schooling (Cairns et al., 1989).

3.6.3 How family influences academic performance

The nature of the relationship between home and school- microsystems has been examined for decades using a unidirectional model, wherein home environment characteristics are analysed

for their effects on school-related outcomes. Because of the abundance of work in this area, scholars of economics, sociology, education, and psychology now largely agree that parental education, socioeconomic status, and family structure have main effects on the academic performance of youth (Reardon, 2011) and an aggregated effect on system-level school performance indicators, such as state-mandated standardized test scores (Bryk, Sebring, Allensworth, Luppescu, & Easton, 2010). In other words, students who experience more family related advantages and resources at home tend to perform better academically, and schools that enrol higher proportions of these students tend to be deemed successful by state accountability standards.

Current theory and a growing body of literature support the basic concept that as family problems increase, the risk for a series of negative outcomes substantially increases for children (Brooks-Gunn, Klebanov, Liaw, & Duncan, 1995). Negative outcomes for these children can range from behaviour problems and poor academic achievement to drug use, psychiatric disorders and suicide (Myers, Taylor, Arrington, & Richardson, 1992). Regardless of the source of negative child outcomes, there is a body of research supporting the concept of resilience in children. Resilience represents the ability of children to exhibit competencies and successes despite the presence of stressful life events or circumstances (Werner, 1995). Furthermore, research has supported the notion that resilience can be enhanced by promoting positive family functioning, supporting family strengths, and identifying and intervening in family problems (Lester, McGrath, Garcia-Coll, Brem, Sullivan, & Mattis, 1995).

A successful transition into school which supports early school achievement is a very important condition of further success for children. Children who enter school and perform poorly from the beginning, academically or socially, tend to continue having negative experiences in school (Gillespie, 2014). However, although the mechanisms by which early school competence

contributes to positive developmental outcomes are still being investigated, it seems that whatever the mechanism, early school performance tends to predict later success (Durlak, 1997).

Academic success during middle childhood is critical to successful developmental trajectory through this period into adolescence precisely because academic success is critical to developing a healthy, positive view of one's competence and a positive motivational orientation toward learning in school (Dryfoos, 1994). Research has indicated that school-family-community partnerships improve school programs and school climate, increase parents' skills and leadership, connect families with others in the school and the community, and improve children's chances of success in school and life (Epstein, 1995; Henderson & Mapp, 2002)

3.7 Conclusion

Research on the literature of middle childhood regards this as a unique developmental phase in life which constitutes many challenges. Successful experiences in a wide range of settings help to give a child a healthy, positive view of his or her competence, and a positive attitude toward learning and engagement in life's activities and challenges. Family resilience has one of the main effects on the academic performance of learners in middle childhood. Bearing in mind how important successful experiences can be to children of these ages may help the leaders and staff of out-of-school programs to maximize the benefits their programs provide. The following chapter provides insight into the methodology used in this study.

CHAPTER 4

METHODOLOGY

4.1 Introduction

This chapter outlines the methodology used to conduct the present study to accomplish the specific aims and objectives as described in Chapter 1. A detailed explanation is provided for the research design. It also provides an explanation of how the sample was recruited and presents a full description of the sample. Further, this chapter presents the instruments used to collect the data for analysis and a review of their psychometric properties. It also presents the pilot study conducted before undertaking the main study. The data collection and analysis procedures are also discussed. Lastly, the ethical considerations are discussed.

4.2 Aim and objectives of the study

UNIVERSITY of the WESTERN CAPE

Aim of the study

The aim of this study was to examine the relationship between perceived family resilience and the academic performance of learners in the phase of middle childhood.

The objectives of this study were to:

- Determine the perception of family resilience of learners in the phase of middle childhood;
- Assess the academic performance of learners in the phase of middle childhood;
- Examine the relationship between and predictive effects of family resilience and academic performance of learners in the phase of middle childhood.

4.3 Methodological approach

Overall this study utilised a quantitative methodology in order to investigate the nature of the relationship between family resilience (and its factors) and academic performance of children in middle childhood. Quantitative research, according to Creswell (2008), is an inquiry approach, useful for describing trends and explaining the relationship among variables. In quantitative research, the researcher decides what to study; asks specific, narrow questions; collects quantifiable data from participants; analyzes these numbers using statistics; and conducts the inquiry in an unbiased, objective manner (Edmonds & Kennedy, 2010). The reason for this approach is that it suited the nature of the data collection amongst the learners at the various schools. The quantitative research enabled the researcher to assemble all the info relating to family resilience, as well as the data relating to those students' academic performance and was able to analyze those numbers into statistics.

4.4 Research design

A research design is the plan and procedure for research that spans the decisions, from broad assumptions to detailed methods of data collection and analysis. The selection of a research design is also based on the nature of the research problems, the researcher's personal experiences and the audiences for the study (Creswell, 2008). This study used a cross-sectional, correlational research design. A cross-sectional study observes a phenomenon at a particular period of time (Gray, 2009), and one advantage thereof is that it can reveal associations among variables. Correlation research addresses questions about the relationship between two or more variables and the extent to which they co-vary (Arthur, Waring, Coe & Hedges, 2012). When an association is measured numerically, a correlation coefficient is obtained that gives the strength and direction of the relationship between these variables (Gray, 2009). Research variables are concepts that vary in ways that can be observed, recorded and

measured (Miley, O'Melia & DuBois, 2009). Using the cross-sectional, correlation research design allowed the researcher to reveal the associations between these variables.

The aim of this study was to examine the relationship between perceived family resilience and the academic performance of learners in the phase of middle childhood. Thus a correlational design was appropriate as there were scores (derived by answers on questionnaires) on these variables from the same participants and statistical relationships were examined. This design is also cross-sectional because a subset of participants from the population in the Metro Central District was selected at one point in time (Salkind, 2010).

4.5 Population and sample

Sampling can be defined as selecting the elements to be observed (Babbie & Mouton, 2010). The type of sampling which was used in this study was convenient sampling. Convenience sampling includes participants who are readily available and agree to participate in a study (Fink, 1995). Three primary schools were conveniently selected, as they were geographically within easy reach of the researcher.

The population refers to the group (of people) about whom researchers want to draw conclusions (Babbie & Mouton, 2010). This study focused on the Metro Central District within the Western Cape Education Department, which comprises of 68 primary schools (Grade 1 – Grade 7). The sampling frame for this research was Grade 6 learners from the 2015 registered list of learners of the three primary schools enlisted within the Penlyn Estate area. Together, these schools have approximately 285 learners. The population of this research study comprised of 194 Grade 6-learners. The chosen area was geographically convenient for the researcher and the reason for this category of learners is that their ages spanned the phase of middle childhood. Class lists of all the learners at the 3 schools were requested, so that the

researcher could keep track of which learners participated in the actual study, and also for the purpose of measuring their academic performance. All Grade 6 learners were provided with information sheets and consent forms for parents, as well as information sheets and assent forms for participants. Their class teachers explained the purpose of the study to them and they were requested to ask permission from parents/guardians to participate in the study. The learners who obtained written permission from parents participated in the study.

4.6 Data collection instrument

Structured questionnaires were given to the participants to complete. The instrument that were used was: The Family Resilience Assessment Scale (FRAS), together with a section for completion of demographics. The questionnaires were only printed in English.

4.6.1. The Family Resilience Assessment Scale

Family Resilience is conceptualized as behavioural patterns and functional competencies that help families negotiate and cope with crisis and hardships (McCubbin et al., 1998). For this study family resilience serves as an independent variable. To measure family resilience, this study utilized the Family Resilience Assessment Scale (FRAS) developed by Tucker Sixbey (2005) using Walsh's (2006) theoretical framework of family resilience. The measure uses a 4-point Likert Scale that ranges from strongly disagree to strongly agree. The FRAS contains six subscales. The overall internal consistency of the FRAS according to Tucker is $\alpha = 0.96$.

4.6.2 Academic performance measurement

Permission for obtaining the 2015 mid-year results of the participants were requested by the various schools and principals granted the researcher access to this information. Only the researcher and supervisors had access to the names and results of each student. Academic

performance was measured utilising the results provided by the teachers of the participating learners. The mean of the learning areas covered in Grade 6 (English, Afrikaans, Mathematics, Natural Science, History, Geography, Combined Social Science, Life Skills) were calculated and each participant's results were recorded. This was then correlated with the scores obtained on the Family Resilience Assessment Scale questionnaire using the Statistical Package for the Social Sciences (SPSS) V23.

4.7 Data collection

The research was conducted after receiving permission to conduct the study from the University of the Western Cape. Further permission was sought for at the Western Cape Education Department (WCED). The principals of the various selected schools were contacted to get permission to inform parents of participants of the study and to administer questionnaires. Letters were sent home with the children explaining the purpose, aims and objectives of the study (see Appendices III & IV) as well as consent form to be signed (see Appendices V & VI). A date and time for the administration of the main study by the researcher were arranged with the principals of the respective schools. The researcher proceeded to administer the questionnaires at the located venues (viz. school classrooms that were provided). The purpose for the study was stipulated on a cover sheet, followed by a demographics page, which had to be completed. These were provided by the researcher and the questionnaire procedure was carefully explained to all 194 participants respectively (classes of about 35 – 40 learners at a time completed the study). The completed questionnaires were then collected. It was re-iterated in all correspondence that participation is voluntary and that all information shared would remain confidential.

4.7.1 Pilot study

The term pilot study is used in two different ways in social science research. It can refer to so-called feasibility studies which are "small scale version[s], or trial run[s], done in preparation for the major study" (Polit et al., 2001: 467). However, a pilot study can also be the pre-testing or 'trying out' of a particular research instrument (Baker 1994: 182-3). One of the advantages of conducting a pilot study is that it might give advance warning about where the main research project could fail, where research protocols may not be followed, or whether proposed methods or instruments are inappropriate or too complicated. In the words of De Vaus (1993: 54) "Do not take the risk. Pilot test first."

The purpose of a pilot study was to understand from a qualitative perspective if children were able to understand the Family Resilience Assessment Scale (FRAS) because it was not developed with children in mind. Other factors which emerged were: time taken for the participants to complete the questionnaire, language understanding and comprehension of questions. Changes were made because the learners did not understand the level of some of the vocabulary used in the questionnaire. Together with my supervisor we reviewed the problem areas and changed the wording to make it more suitable for children to understand.

Thereafter, telephonic contact was established with the principals of the schools. A date and time for the meeting with the principal to inform them of the study was set up. Information sheets (see Appendix IV) and consent forms (see Appendix VI) that needed to be completed by the parents regarding the study, were left with the principals of the respective schools. Once the consent forms had been collected by the respective teachers, a date and time was set as to when the data collection would be conducted at the particular schools.

The questionnaire was administered to a group that is similar in age and background to the sample. Twenty-seven middle childhood male and female learners from one school was used

in a pilot study to understand from a qualitative perspective if children could understand the Family Resilience Assessment Scale. It was also used to establish any challenges or limitations that may occur with this age group. The participants were informed of the study and were allowed to ask questions or concerns that they had. The learners were given consent forms (see Appendix V) that needed to be signed before administering the questionnaires, if they volunteered to partake in the study. Learners completed the questionnaires within 20-30 minutes.

4.7.1.1 Challenges identified during the pilot study

During the pilot study it was evident that some of the wording was too complex for middle childhood learners to comprehend. In the 'demographics' section, some learners did not understand the meaning of 'siblings' and that had to be simplified by saying 'brothers and sisters'. In spite of the questionnaire's being read through to the learners, a number of the questions still needed explanation—such as: "We can blow off steam at home without upsetting someone". It had to be explained that this meant expressing themselves (participants i.e.) in whichever way they felt was appropriate at any given time, and that their behaviour would not offend anyone within the family (see Table 4.1).

4.7.1.2 Changes made to the instrument

Because of difficulties encountered, certain questions in the Family Resilience Assessment Scale had to be reviewed so that learners could comprehend its meaning. The following changes were then made:

Table 4.1 Changes made to the measuring instrument

FROM	<u>TO</u>
1. Our family structure is flexible to deal with the unexpected.	Our family is flexible to deal with unexpected events eg. moving home.
5. We accept that problems occur unexpectedly.	We accept that problems which we do not expect, can happen.
8. We are adaptable to demands placed on us as a family.	We are flexible/able to cope with demands placed on us as a family.
14. We can ask for clarification if we do not understand each other.	We can ask for an explanation if we do not understand each other.
16. We can blow off steam at home without upsetting someone.	We can show our feelings at home without upsetting someone.
17. We can compromise when problems come up.	We can "make a deal" with each other when problems come up.
25. We consult with each other about decisions.	We talk to each other about decisions we need to make.
33. We feel taken for granted by family members.	We feel taken advantage of by family members.
42. We participate in church activities.	We take part in church/mosque activities.
44. We seek advice from religious advisors.	We seek advice from religious advisors, such as priests, imams etc.
53. We understand communication from other family members.	We understand how other family members talk to each other.

The table above was extracted from the original Family Resilience Assessment Scale (Sixbey, 2005), which consists of 54 items. This scale was originally intended for adult participants. The items listed were reviewed and restructured to make meanings easier to comprehend for the participants of the main study. Please see the revised questionnaire attached (see Appendix II).

4.8 Data collection for the main study

Collecting data, according to Creswell (2009), means identifying and selecting participants for a study, obtaining their permission to conduct the study and gathering information by asking questions or observing behaviour. Once the pilot study was completed, i.e. both the data collection and the analyses of the research process and the questionnaire, the results of this process informed the process for the main study. The data for the main study was collected at the same and different schools to those of the pilot study. As with the pilot study, the same process of requesting permission was followed: participants were informed of the study and were allowed to ask questions or concerns that they had. The learners were given consent forms (see Appendix V) that needed to be signed before administering the questionnaires, if they volunteered to partake in the study. Participants, their parents and teachers were informed about the study. The participants were given the opportunity to decide on voluntarily participating in the study, after which the questionnaires were dispensed to groups of learners. As with the pilot study, these groups consisted of 30 to 40 learners at a time, in a class room setting within the limitations of the school time table. Completion of the questionnaires took approximately 20 - 30 minutes.

4.9 Data analysis

Analysing and interpreting data involves drawing conclusions from the data collected and presenting the findings in tables, figures and pictures, to verify the conclusions and provide answers to the researcher's research questions (Creswell, 2009). Descriptive statistics are used to describe or summarize a set of data, while inferential statistics are used to make inferences from the sample chosen to a larger population (Gray, 2009). The key variables in this research study were the relationship between family resilience and academic performance of learners in the phase of middle childhood.

The raw data collected from the research conducted was entered into the Statistical Package for the Social Science (SPSS) V23. The data was coded, cleaned, checked for errors and analysed using descriptive and inferential statistics. Descriptive statistics include frequencies and means, while inferential statistics include a Pearson correlations and linear regression analysis. Regression analysis is a quantitative research method that is used when the study involves modelling and analysing several variables, where the relationship includes a dependent variable and one or more independent variables to predict effects (McGrath, 2011).

4.10 Reliability and validity

The test-retest method was implemented in the pilot study, which assisted in measuring the reliability of the instrument. Reliability is an indication of consistency between two measures of the same thing (Black, 1999, as cited in Gray, 2009). For the research tool to be classified as reliable, it is expected to provide the same results when something was measured at different times.

WESTERN CAPE

Validity applies to particular interpretations of tests, assessments, questionnaires or other data collection instruments (Arthur et al., 2012). According to David (2010), it is the degree to which whatever is measured is what the researcher intended. To ensure validity, a research instrument must measure whatever it was intended to measure. To achieve validity, the research instrument subject area and the operationally defined subject areas must match exactly (Gray, 2009). The Family Resilience Assessment Scale has been used. The Cronbach Alpha scores of this research for the various subscales were as follows: Family Communication and Problem Solving .90; Utilizing Social and Economic Resources .77; Maintaining Positive Outlook .71; Ability of Make Meaning of Adversity .37; Family Spirituality .65; Family Connectedness .28.

4.11 Ethics statement

Permission to conduct the study was requested from the Ethics Committee of the University of the Western Cape. Once ethical clearance was received, permission to collect the data was requested from the Western Cape Department of Education in order to gain access to the various schools. The following ethical guidelines were followed in the pilot study and the main study:

- (i) The purpose of the study was explained to the participants and their parents (see Appendix III & IV);
- (ii) Written Informed consent was requested from the parents of the participants (see Appendix III) and Assent forms were completed by the participants before participation (see Appendix IV);
- (iii) participation in the study was voluntary;
- (iv) participants were informed of their right to withdraw from the study, at any time,without prejudice;
- (v) participants were treated with respect and dignity;
- (vi) all questionnaires were coded with numbers to ensure anonymity of participants;
- (vii) all information was handled confidentially and only used for the benefit of the study;
- (viii) the questionnaires were collected and kept in a secure place;
- (ix) there was no known risks involved with participation in the study; and

(x) the research findings and feedback were disseminated to the various schools that assisted in the research study

4.12 Conclusion

The chapter provided the methodological design of the study. A cross-sectional correlational design was used to achieve the main aim and objectives of the study. More specifically, the design used was a systematic investigation of relationships among family resilience and academic performance in the phase of middle childhood. The chapter provided information with regard to the various stages of the research process such as sampling, data collection and data analysis. This chapter has outlined the research pilot study, the main study, and the changes made as a result of the pilot study. The results of the research follow in the next chapter.



CHAPTER 5

RESULTS

5.1 Introduction

This chapter presents the results of the analysis for this study. The analysis was conducted using the Statistical Package for the Social Sciences (SPSS) V23. This chapter presents the results as (1) descriptive information regarding knowledge of family resilience and academic performance of learners in the phase of middle childhood, (2) the relationships between and the predictive effects of family resilience and academic performance of learners in the phase of middle childhood.

The following is a coding guide used to analyze the data:

Abbreviation	Variable
FCPS	WESTErFamily communication and problem
	solving
USER	Utilizing social and economic resources
MPO	Maintaining positive outlook
AMMA	Ability to make meaning of adversity
FS	Family spirituality
FC	Family connectedness

5.2 Overview

The overview of the analysis lies within the aims and objectives given below:

5.2.1 Aim of the study

The aim of this study is to examine the relationship between perceived family resilience and the academic performance of learners in the phase of middle childhood.

5.2.2 Objectives

The objectives of the study are to:

- Determine the perception of family resilience of learners in the phase of middle childhood;
- Assess the academic performance of learners in the phase of middle childhood;
- Examine the relationship between and predictive effects of family resilience and academic performance of learners in the phase of middle childhood.

The hypothesis of the study based on the aims and objectives in Chapter 1, proposed that there is a significant predictive relationship between family resilience (family communication and problem-solving, maintaining a positive outlook, making meaning of adversity, family spirituality, family connectedness and utilising social and economic resources) and learners' academic performance in the phase of middle childhood.

5.3 Internal consistency

The study used one instrument to measure the variables under study. This was the Family Resilience Assessment Scale (FRAS) as developed by Sixbey (2005). The FRAS was designed

as a family measure for administration with one adult family member. However, in this study, it was restructured to be administered to children. The FRAS was adapted on the basis of the pilot study. The internal consistency is indicated by Cronbach alphas in Table 5.1.

Table 5.1 Internal consistency

Variables	Alpha
Family communication and problem solving	.90
Utilizing social and economic resources	.77
Maintaining positive outlook	.71
Ability to make meaning of adversity	.37
Family spirituality	.65
Family connectedness	.28
Total Scale	.91

Cronbach Alpha coefficients above .75 are deemed to be acceptable, while .6 is considered to be moderately acceptable (Anastasi, 1982). The alphas show a good reliability of the instruments used to measure the variables. Cronbach Alpha coefficients, Family Communication and Problem Solving (.90) and Utilizing social and economic resources (.77) showed good coefficients, whilst ability to make meaning of adversity (.37) and Family connectedness (.28) did not show good coefficients.

Table 5.2 Demographic information of participants

Cable 5.2	Demographic information of p	articipants	N 104	0/
	Variables		N = 194	%
				_
		Male	72	37.1
	Gender	Female	122	62.9
		11	77	39.7
		12	104	53.6
		13	12	6.2
	Age	14	1	.5
	Grade	VI	194	100
	Grauc	V1	1)4	100
		Coloured	189	97.4
	11-11-11	Black/African	1	.5
		White		
	Race	Indian	4	2.1
	UNIVERS	SITY of the		
	WESTER	Afrikaans	80	41.2
		English	114	58.8
		isiXhosa		
	Home language	Other		
		Mother	176	90.7
		Father	142	73.2
		Grandparent	78	40.2
		Uncle/Aunt	52	26.8
	I live with	Sibling	161	83
Is thoras	child with a disability living in	Yes	6	3.1
18 mere a		No	188	96.9
	your home?	140	100	90.9

Table 5.2 presents an overview of the demographic profile of the 194 participants in this study. Demographic information includes gender, age, grade, race, home-language, "I live with" and "Is there a child with disability living in your home"?

The results show that the majority of the participants were female [122 (62.9%)]. All the participants were in Grade VI. Of the participants 189 (97.4%) identified themselves (according to terminology used by the previous government to separate individuals) as "Coloured", 1 (.5%) as "Xhosa" and 4 (2%) as "Indian". No "Whites" participated in the study. There were 114 (58.8%) English-speaking participants and 80 (41.2%) were Afrikaansspeaking. The majority of the participants 176 (90.7%) live with their mother, whilst 142 (73.2%) live with their father, 78 (40.2%) live with their grandparent, [52 (26.8%)] live with an aunt/uncle and 161 (83%) live with siblings. Only [6 (3.1%)] participants live with a child who has a disability and 188 (96.9%) do not have anyone with a disability living with them.

5.4 Perception of family resilience

Resilience - the ability to withstand and rebound from disruptive life challenges - has become an important concept in mental health theory, research, and practice over recent decades. It involves dynamic processes fostering positive adaptation within the context of significant adversity (Bonanno, 2004; Luthar, 2006; Masten, 2001).

WESTERN CAPE

This section presents the Means (M) and Standard Deviations (SD) of the subscales of the Family Resilience Assessment Scale, considered the most important dimensions to determine family resilience.

Table 5.3 Subscale on Family Communication and Problem Solving (FCPS)

Item	N	M	SD
1 Our family is flexible to deal with unexpected events eg. moving home.	194	3.07	.88
6 We all have input into major family decisions	194	3.03	.79
7 We are able to work through pain and come to an understanding	194	3.23	.80
8 We are flexible/able to cope with demands placed on us as a family.	194	3.11	.76
9 We are open to new ways of doing things in our family	194	3.25	.67
We are understood by other family members	194	3.36	.71
We can ask for an explanation if we do not understand each other.	194	3.31	.70
We can be honest and direct with each other in our family	194	3.29	.77
We can show our feelings at home without upsetting someone.	194	2.92	.87
We can "make a deal" with each other when problems come up.	194	3.22	.66
We can deal with family differences in accepting a loss	194	3.20	.79
We can question the meaning behind messages in our family	194	3.21	.73
23 We can talk about the way we communicate in our family	194	3.08	.83
We can work through difficulties as a family	194	3.18	.78
We talk to each other about decisions we need to make.	194	3.22	.73
26 We define problems positively to solve them	194	3.19	.61
We discuss problems and feel good about the solutions	194	3.26	.67

28	We discuss things until we reach a solution	194	3.22	.72
29	We feel free to express our opinions	194	3.19	.78
30	We feel good giving time and energy to our family	194	3.41	.71
40	We learn from each other's mistakes	194	3.44	.59
41	We mean what we say to each other in our family	194	3.22	.76
46	We share responsibility in the family	194	3.47	.64
48	We tell each other how much we care for one another	194	3.29	.73
52	We try new ways of working with problems	194	3.25	.68
53 each o	We understand how other family members talk to other.	194	3.18	.72
54 emotio	We work to make sure family members are not onally or physically hurt	194	3.40	.79

Average total percentage scores for the Family Resilience Assessment Scale (N=194)

Responses were on a Likert scale of Strongly Agree = 4, Agree=3, Disagree=2, Strongly Disagree=1

Table 5.3 illustrates the scoring of Family Communication and Problem Solving. The majority of participants indicated the following as the most prevalent items: 'We share responsibility in the family' (M = 3.47, SD = .64); this was followed by 'We learn from each other's mistakes' (M = 3.44, SD = .59); the least prevalent responses were: 'We all have input into major family decisions' (M = 3.03, SD = .79) and 'We can show our feelings at home without upsetting someone' (M = 2.92, SD = .87).

In terms of the aim and objectives of the current study, it is evident from the table above that families that work together responsibly in accomplishing tasks and where there are communication aspects and learning from each other, are able to create a sense of academic achievement within the learner in the phase of middle childhood.

Table 5.4 Utilizing Social and Economic Resources (USER)

	Item	n	M	SD
11	We ask neighbours for help and assistance	194	2.90	.90
19	We can depend upon people in this community	194	2.69	.97
31 emerg	We feel people in this community are willing to help in an ency	194	3.09	.85
32	We feel secure living in this community	194	2.76	1.1
38	We know there is community help if there is trouble	194	2.99	.90
39	We know we are important to our friends	194	3.34	.75
43	We receive gifts and favours from neighbours	194	3.05	.90
49	We think this is a good community to raise children	194	2.72	1.1

Table 5.4 presents the scoring of Utilizing Social and Economic Resources. The majority of participants indicated 'We know we are important to our friends' (M = 3.34, SD = .75); following that was 'We feel people in this community are willing to help in an emergency' (M = 3.09, SD = .85), indicating the highest mean-scores. These were the responses on the last two items, showing the lowest mean-scores: 'We think this is a good community to raise children' (M = 2.72, SD = 1.1); and 'We can depend upon people in this community' (M = 2.69, SD = .97). Whilst families are able to thrive socially amongst themselves and work out problems within the home, problems within the community are likely to cause difficulties.

Table 5.5 Maintaining a Positive Outlook (MPO)

	Item			SD
13	We believe we can handle our problems	194	3.24	.69
21	We can solve major problems	194	3.16	.78
22	We can survive if another problem comes up	194	3.04	.87
34	We feel we are strong in facing big problems	194	3.21	.77
36	We have the strength to solve our problems	194	3.29	.73
51	We trust things will work out even in difficult times	194	3.38	.73

Table 5.5 presents the scoring for Maintaining a Positive Outlook. Most participants had prevalent responses in the following two items by saying 'We trust things will work out even in difficult times' (M = 3.38, SD = .73); following that item was 'We have the strength to solve our problems' (M = 3.29, SD = .73); and least prevalent responses were in the following two items: 'We can solve major problems' (M = 3.16, SD = .78); and finally 'We can survive if another problem comes up' (M = 3.04, SD = .87).

Table 5.6 Family Connectedness

	Item	n	M	SD
2	Our friends value us and who we are	194	3.36	.66
33	We feel taken advantage of by family members.*	194	2.13	.97
37	We keep our feelings to ourselves *	194	2.61	.89
45 *	We seldom listen to family members' concerns or problems	194	2.51	.93
47	We show love and affection for family members	194	3.59	.66
50 comm	We think we should not get too involved with people in unity *	194	2.71	.96

Table 5.6 presents the scoring for Family Connectedness. Most prevalent responses were: 'We show love and affection for family members' (M = 3.59, SD = .66); this was followed by 'Our friends value us and who we are' (M = 3.36, SD = .66); the final two items were responded least prevalently in the following manner: 'We seldom listen to family members' concerns or problems' (M = 2.51, SD = .93) and 'We feel taken advantage of by family members' (M = 2.13, SD = .97).

Table 5.7 Family Spirituality (FS)

	Item	n	M	SD
12	We attend church/synagogue/mosque services	194	3.61	.60
35	We have faith in a supreme being	194	3.67	.55
42	We take part in church/mosque activities.	194	3.51	.71
44 imams	We seek advice from religious advisors, such as priests, setc.	194	3.35	.75

Table 5.7 presents the scoring for Family Spirituality. The following items had the most responses: 'We have faith in a supreme being' (M = 3.35, SD = .75); this was followed by 'We attend church/synagogue/mosque services' (M = 3.61, SD = .60); then came 'We take part in church/mosque activities' (M = 3.51, SD = .71); finally, 'We seek advice from religious advisors, such as priests, imams etc.' (M = 3.35, SD = .75).

Table 5.8 Ability to Make Meaning of Adversity (AMMA)

Item	n	M	SD
3 The things we do for each other make us feel a part of the family	ne 194	3.34	.72
4 We accept stressful events as a part of life	194	3.16	.76
5 We accept that problems which we do not expect, can happen.	an 194	3.25	.72

Table 5.8 presents the scoring for Ability to Make Meaning of Adversity. The most prevalent responses were to the following items: 'The things we do for each other make us feel part of the family' (M = 3.34, SD = .72); this was followed by 'We accept that problems which we do

not expect, can happen' (M = 3.25, SD = .72); and lastly: 'We accept stressful events as a part of life' (M = 3.16, SD = .76).

Table 5.9 Table of the mean scores for the variables of family resilience

Descriptive statistics							
	N	Minimum	Maximum	Mean	Standard Deviation		
FCPS	194	1.63	4.04	3.23	.39		
USER	194	1.38	4.00	2.94	.58		
MPO	194	1.67	4.00	3.12	.48		
AMMA	194	2.00	4.00	3.26	.49		
FS	194	2.25	4.00	3.54	.46		
FC	194	1.83	4.00	2.83	.45		
Academic Performance	194	13.86	95.29	62.00	15.05		

Table 5.9 presents the scoring for the variables of family resilience. The following items had the most prevalent responses: 'Family Spirituality' (M = 3.54, SD = .46); this was followed by 'Ability to Make Meaning of Adversity' (M = 3.26, SD = .49). The least prevalent responses were in the following two items: 'Utilizing Social and Economic Resources' (M = 2.94, SD = .58) and 'Family Connectedness' (M = 2.83, SD = .45)

5.5 Academic Achievement

This study is based on existing data received from the schools where research was conducted. Provision was made for the 2015 midyear results of all the participants in the study (N = 194).

Table 5.10 Percentage scores for the Academic achievement for the total sample (N=194).

Variables	N	Minimum	Maximum	M	SD
English	194	16	93	65.97	12.79
Afrikaans	194	6	94	58.54	18.35
Math	194	8	95	51.92	19.54
Natural Science	194	14	98	62.88	19.15
History	194	5	100	66.66	19.04
Geography	194	5	100	61.05	18.83
Combined Social Science	194	5	100	64.29	17.27
Life Skills	194	27	88	66.98	12.58
Academic performance	194	14	95	62.00	15.05

The results in Table 5.10 present the average scores for Academic Achievement during Midyear results of the 2015 academic school year. The results indicate that participants scored the highest for Life Skills (M = 66.98, SD = 12.58); this was followed by History (M = 66.66, SD = 19.04); English (M = 65.97, SD = 12.79); Combined Social Sciences (M = 64.29, SD = 17.27); Natural Science (M = 62.88, SD = 19.15); Afrikaans (M = 58.54, SD = 18.35); and the lowest scores were attained for Mathematics (M = 51.92, SD = 19.54)

5.6 The relationship between the variables

This section provides the results of the relationship between the variables. A correlation (Table 5.11) was first conducted to determine the relationship between the subscales of family

resilience. This correlation was then followed by separate regression analyses (Table 5.12) to determine whether there is a relationship between family resilience and academic performance.

 Table 5.11
 Relationship between family resilience and academic performance

100010	23.11 Kelationship be	***************************************			пастие ре		Ĭ
Var	riables	1	2	3	4	5	6
1	Family	-					
1	Communication &						
	Problem Solving						
_	Utilizing Social &	.48**					
2	Economic Resources						
2	Maintaining a Positive	.74**	.56**				
3	Outlook						
4	Ability to Make	.61**	.21**	.51**			
	Meaning of Adversity						
5	Family Spirituality	.54**	.23**	.35**	.26**		
	- II G	**	**	0.**	2.2**	• 0 **	
6	Family Connectedness	.42**	.21**	.28**	.32**	.20**	
7	Academic Performance	.31**	.29**	.41**	.37**	.24**	.39**
7							

^{**}Correlation is significant at the 0.01 level (2-tailed)

The results in Table: 5.11 show that there is a significant positive relationship between academic performance and family resilience (r = .31, p = < 0.01). There were significant positive relationships to be found between Academic Performance and (i) Utilising Social and Economic Resources (r = .29, p < 0.01); (ii) Maintaining a Positive Outlook (r = .41, p < 0.01); (iii) Ability to Make Meaning of Adversity (r = .37, p < 0.01); (iv) Family Spirituality (r = .24, p < 0.01); (v) Family Connectedness (r = .39, p < 0.01).

 Table 5.12
 Regression analyses predicting academic performance

Family Resilience	В	SE b	β	T	P
Constant (Academic Performance)	-9.41				
Family Communication and Problem Solving	-13.56	4.39	35	-3.09	.002*
Utilizing Social and Economic Resources	3.34	1.98	.13	1.68	.09
Maintaining a Positive Outlook	10.28	3.07	.33	3.35	.001*
Ability to Make Meaning of Adversity	7.81	2.43	.25	3.21	.002*
Family Spirituality	4.97	2.38	.15	2.09	.04*
Family Connectedness	10.43	2.26	.31	4.62	.000*

Note Academic Performance: $\Delta R^2 = 0.29$

Table 5.11 presents the results for a linear regression analyses. Academic performance was significantly predicted by (i) Family Communication and Problem Solving ($\beta = -.35$, $p = \le .002$); (ii) Maintaining a Positive Outlook ($\beta = .33$, p = .001); (iii) Ability to Make Meaning of Adversity ($\beta = .25$, p = .002); (iv) Family Connectedness ($\beta = .31$, p = .000); (v) Family Spirituality ($\beta = .15$, p = .04).

The model explains a 29% variance for academic performance.

5.7 Summary of findings

The results of this study show that there is a significant predictive relationship between family resilience (family communication and problem-solving, utilising social and economic resources, maintaining a positive outlook, making meaning of adversity, family spirituality, family connectedness) and learners' academic performance in the phase of middle childhood. The most prevalent correlation was found in the subscales of Maintaining a Positive Outlook and Family Connectedness. This was followed by Ability to Make Meaning of Adversity and

Family Communication & Problem Solving. The least prevalent subscales in the relationship between academic performance and family resilience were on the subscales of Utilizing Social & Economic Resources and Family Spirituality.



CHAPTER 6

DISCUSSION, CONCLUSION AND RECOMMENDATIONS

6.1 Introduction

The current chapter is an investigation of the relationship between family resilience and student academic performance in the phase of middle childhood. This chapter presents a discussion of the findings of the study. The findings in Chapters 4 and 5 are examined in relation to the aims and hypotheses of the thesis discussed in Chapter 1 and integrating the conceptual framework discussed in Chapter 2. This chapter also elaborates on family resilience, academic achievement and middle childhood. Finally, the limitations of the study are discussed and recommendations offered.

6.2 Prevalence of family resilience in accordance with the subscales of The Family Resilience Assessment Scale

The summary of the findings based on the objectives of the study is that there is a significant predictive relationship between family resilience and learners' academic performance in the phase of middle childhood. It is important to bear in mind that there is not much literature available on the prevalence of family resilience in the South African context, but internationally, there has been considerable research in this field.

Family resiliency is the ability of a family to respond positively to an adverse event and to emerge strengthened, more resourceful and more confident (Simon et al., 2005). Resilience develops not through the evasion of adverse events, but through a family's successful use of protective factors to cope with these events and become stronger (O'Leary 1998). Given that families are diverse and reside in dynamic environments, it is assumed that family resilience

varies over time, and is a process rather than an outcome (Rutter 1999). One can also argue that the presence of resilience is only possible because of the experience of adversity. Resiliency is fostered by protective factors and inhibited and developed owing to risk factors. Protective factors modify or transform responses to adverse events so that families avoid possible negative outcomes. Conversely, risk factors are circumstances that increase the probability of poor outcomes. Protective and risk factors are not static entities; they change in relation to context (Walsh 2003), which leads to different outcomes.

Silliman (1994), Masten (2001) and Rutter (1990) are amongst many family scientists, who are explaining the phenomenon of rising above challenges by the strength-based model of resilience; examining positive coping factors rather than deficits. Consistent with a strengths approach and adaptability, McCubbin and McCubbin (1988) defined family resilience as "charateristics, dimensions, and properties of families which help families to be resilient to disruption in the face of change and adaptive in the face of crisis situations" (p. 247). It follows that healthy (strong) families may be explained as sharing resilient processes in response to stress or change. Family resilience clarifies why some families not only survive problems but thrive and become stronger as a result. The qualities of resilience can be developed at any time of the family life cycle and are often enhanced and refined through problem solving (Walsh, 1998, 2003b). Family interactions and a general optimistic attitude are at the heart of weaving a strong and unique tapestry of resilience as families continue to love and support each other through normative and unexpected challenges.

The results of this study are based on the nine factor structure supporting Walsh's (1988) model of family resilience, which Sixbey then transformed into six factors based on her scale. The preliminary factor analysis was not conducted on the pilot sample due to the small number of participants (n = 27). However, changes were made from the pilot participant feedback and

the instrument was then completed by 194 participants. The dimensions of the Family Resilience Assessment Scale, which include Family Communication and Problem-solving (FCPS), Utilizing Social and Economic Resources (USER), Maintaining a Positive Outlook (MPO), Family Connectedness (FC), Family Spirituality (FS) and Ability to Make Meaning of Adversity (AMMA), are processes which make up family resilience.

6.2.1 Family Communication and Problem Solving (FCPS):

The results of the current study indicated the following items were most prevalent in the subscale of FCPS: 'We learn from each other's mistakes' and 'We share responsibility in the family'. The least prevalent items were 'We are able to work through pain and come to an understanding' and: 'We can show our feelings at home without upsetting someone.'

According to Walsh (1998b) communication processes that clarify ambiguous situations, encourage open emotional expression and empathic response, and foster collaborative problem solving are especially important in facilitating resilience. Research suggest that indicators of quality time with one's family (e.g., regular family meals; Eisenberg et al. 2004) and certain after-school activities (e.g., sports, music) are positively related to children's well-being (Bastian, 2000; Donaldson and Ronan, 2006; Posner and Vandell, 1999; Steptoe and Butler, 1996). It is in forums of this nature that family communication and problem solving are enhanced. The interactivity of these family times creates a sense of belonging for the middle childhood learner, who in turn feels secure that problems will be addressed and possible solutions to them will materialise.

6.2.2 Utilizing Social and Economic Resources (USER)

Most prevalent items in this subscale were found to be: 'We know we are important to our friends' and 'We feel people in this community are willing to help in an emergency'. The items

that were least prevalent were: 'We think this is a good community to raise children' and: 'We can depend upon people in this community'.

Previous research by Staveteig and Wigton (2005) state that all families benefit from community resources, but those with a host of other social problems can particularly benefit from external support. According to Luthar (1999) the poor have all the personal and family problems everyone else has, along with complications of poverty, isolation, and a loss of community. Family isolation with a lack of social support can erode resilience, particularly under stressful conditions.

Children who perceive their community and neighbourhood as safe and satisfying have been found to benefit through improved academic grades and fewer problem behaviours than those who see their neighbourhood as threatening (Woolley & Grogan-Kaylor, 2006). Particularly for families living in impoverished or isolated conditions, the availability and use of quality support systems can dramatically increase positive outcomes, such as perseverance, hope, education, and companionship (Orthner et al., 2004). If parents are not able to provide emotional support for children, other kin or mentors may step in to offer the children support, positive role models, and protection (Aronowitz & Morrison-Beedy, 2004). Healthy families have the strength to admit when they need help and are more likely to turn to extended family, friends, neighbours, community services, or counselling (Walsh, 1998). Unfortunately, there appear to be two trends diminishing a family's interaction with the community. One is the tendency to withdraw from society; the other is a lack of available services. The trends toward relocation and smaller family size in modern society have frayed kin and community connections, requiring the establishment of new forms of social support (Hanson, 2001a).

6.2.3 Maintaining a Positive Outlook (MPO)

problems' were the two most prevalent items in the subscale of MPO, and 'We can solve major problems' and 'We can survive if another problem comes up' were the least prevalent items.

An optimistic confidence in overcoming odds lies at the heart of resilience. Rutter (1987) identified an approaching style to new situations and a sense of humour as important features in viewing crises as meaningful or comprehensible. The importance of optimism was emphasized again by Rutter in 1993:

'We trust things will work out even in difficult times' and 'We have the strength to solve our

Almost certainly, it is not that there is one "right" way of thinking about things, or one optimal style of coping. Rather, what seems important is to approach life's challenges with a positive frame of mind, a confidence that one can deal with the situation, and a repertoire of approaches. (p. 630)

Bradbury and Karney (2004) reported that displays of positive emotions (such as affection, WESTERN CAPE) humour, offering positive solutions, and accepting suggestions) were more influential in marital satisfaction than problem solving or communication. Hope for the future was found to foster resilience, measured by the avoidance of risky behaviours, among 443 impoverished inner-city black female adolescents and to thwart risk-taking behaviours in Aronowitz and Morrison-Beedy's (2004) study. Affirming strengths and possibilities, maintaining courage and hope, and remaining optimistic have been shown to be effective methods of coping for families (McCreary & Dancy, 2004).

The comfort and security of a resilient home may be seen as a protective and secure base for all family members (Bowlby, 1982). In times of crisis, the family's pulling together and turning to each other for support has been found to be one of the most important recovery factors in resilience (White et al., 2004).

On the contrary, unpreparedness and the inability to tackle or solve major problems, creates a sense of uncertainty and instability within the middle childhood learner and family resilience is stifled in such instances.

6.2.4 Family Connectedness (FC)

The most prevalent items on the subscale of FC were: 'We show love and affection for family members' and 'Our friends value us and who we are'. Least prevalent items on the subscale were: 'We seldom listen to family members' concerns or problems' and 'We feel taken advantage of by family members'.

According to Olson (2000) the premise of the systems approach to families is that the whole and its parts must be connected and content for both to continue. A family system works when its members feel good about the family, needs are being met, and the development of relationships flows smoothly. McCubbin et al. (2002) and Walsh (1998) further state that family cohesion enhances family confidence that problems are comprehensible, manageable, meaningful, and that higher levels of reorganization and adjustment are reached after the crisis. Well-functioning family members tend to interact on a daily basis with nurturance and compliments, and they reinforce each-others' efforts (Walsh, 2003a).

6.2.5 Family Spirituality (FS)

The following items were the most prevalent items in the subscale of FS: 'We have faith in a supreme being' and 'We attend church/synagogue/mosque services'. The least prevalent items were: 'We take part in church/mosque activities' and 'We seek advice from religious advisors, such as priests, imams, etc.'

According to Antonovsky and Sourani (1988) families can be helped to gain a sense of coherence by recasting a crisis as a shared challenge that is comprehensible, manageable, and meaningful to tackle. Normalizing and contextualizing members' distress as natural or understandable in their crisis situation can soften their reactions and reduce blame, shame, and guilt. Drawing out and affirming family strengths in the midst of difficulties helps to counter a sense of helplessness, failure, and despair as it reinforces pride, confidence, and a "can-do" spirit. The encouragement of family members bolsters efforts to take initiative and persevere in efforts to overcome barriers. Although spiritual resources have been largely neglected in clinical practice, they can be tapped as wellsprings for resilience (Walsh, 1999).

6.2.6 Ability to Make Meaning of Adversity (AMMA)

In the subscale of AMMA, the most prevalent items were: 'The things we do for each other make us feel a part of the family' and 'We accept that problems which we do not expect, can happen.' The least prevalent item was: 'We accept stressful events as part of life.'

WESTERN CAPE

Landau (2007) states that strong families are able to maintain relative equilibrium as they manage both the more common day-to-day challenges with those that are more unique to the typical family's experience. According to what researchers have found, Bermudez (2013) highlights a number of family characteristics that promote resilience to change. These resilience-promoting characteristics can be seen throughout the family system. Healthy couples are able to be flexible and adapt as a team to new circumstances in the event of transitions or change (Gottman, 2011). Within the parent-child relationship, healthy parents encourage children's adaptive responses to change, are responsive to children's distress, and access needed social and economic resources that enable the family to maintain equilibrium (Mikulincer & Shaver, 2007). Resilience is vital for individuals and families to be able to deal

with inevitable changes that will happen. Strong families will strive to be resilient in adapting to everyday changes and not just when extreme situations or traumas occur.

6.3 The relationship between family resilience and academic performance of learners in the phase of middle childhood

The skills, attitudes and behaviours of the middle childhood learner factor into their academic success. Other contributing resources are cognitive abilities, beliefs and motivation, with IQ being one of the most powerful predictors of success. Academic competence is also influenced by beliefs and attitudes about school (Stevenson et al. 1993) and motivations to succeed (Henderson & Dweck, 1990).

South African students score at low levels in mathematics and language tests even when compared with students in other African countries (van der Berg and Louw, 2006). Whilst there is widespread agreement that the main challenge in South Africa is the quality of education, there is little empirical analysis that helps policy makers understand the low level of student performance in South African schools or how to improve it.

"A healthy family is neither average nor merely lacking in negative characteristics. Rather, it has described, positive features" (Epstein, Ryan, Bishop, Miller, & Keitner (2003, p. 581). A child's home and school represent distinct microsystems that shape important developmental outcomes. Each microsystem presents its own set of potential risks, or threats to development, as well as protective and promotive factors that counteract those threats and facilitate wellbeing (Masten, Herbers, Cutuli, & Lafavor, 2008). The nature of the relationship between home and school microsystems has been examined for decades using a unidirectional model, wherein home-environment characteristics are analysed for their effects on school-related outcomes. Because of the abundance of work in this area, scholars of economics, sociology, education,

and psychology now largely agree that parental education, socioeconomic status, and family structure have main effects on the academic performance of youth (Coleman et al., 1966; Reardon, 2011).

A family resilience perspective considers each interaction between home and school as an opportunity to strengthen a family's capacity to overcome adversity and successfully rear its children (Walsh, 1998, 2003). Two basic premises guide this resilience theory approach. The first premise is that while stressful crises and persistent economic, physical, and social challenges influence the whole family, key family processes mediate the impact of these crises. A second premise is that while family processes mediate how children are prepared to participate in school, these key family processes can be strengthened by the way the school responds to families. As the family becomes more resourceful, its ability to rear its children is enhanced. As a result, each family-school intervention also can be a preventive measure (Amatea et al., 2006).

The home-environment characteristics viz. family resilience, is essential for positive effects on school-related outcomes. Students who experience more family related advantages and resources at home tend to perform better academically, and schools that enrol higher proportions of these students tend to be deemed successful by state accountability standards.

UNIVERSITY of the

The results of this study indicated that the highest scores in the relationship between family resilience and academic performance were: Maintaining a Positive Outlook (MPO) and Family Connectedness (FC). This was followed by Ability to Make Meaning of Adversity (AMMA) and Family Communication and Problem Solving (FCPS). The lowest scores were found in Utilising Social and Economic Resources (USER) and Family Spirituality (FS).

6.3.1 Maintaining a Positive Outlook (MPO)

Maintaining a positive outlook does not mean that negative behaviours are ignored, but rather that various viewpoints and each member's opinions are examined to identify underlying problems, always augmenting positive aspects (Leon, 2003). A sense of hope, purpose, and meaning in life are vital to emotional well-being, evasion of risk behaviours, and goal-directed behaviours and cognitions (Duke, Borowsky, Pettingell, & McMorris, 2011). There is evidence that children who hold a positive outlook for their future manage stress better than those with less optimism, an indication that hopefulness, as a personal attribute, promotes healthy adaptation and resilience in the face of risk (Wyman, Cowen, Work, & Kerley, 1993). This positive outlook, coupled with a sense of security in the knowledge that their families are able to rebound from challenging situations, are necessary to assist learners to cope with challenges and gear them up for positive academic performance.

6.3.2 Family connectedness (FC)

Resilience research has taken many different forms, used many means of assessment throughout, and shown many different key concepts to be of importance. Despite flaws in early resilience studies, recent studies continue to corroborate the importance of similar factors such as connections to competent and caring adults in the family and community, cognitive and self-regulation skills, positive views of self, and motivation to be effective in the environment (Masten, 2001).

Because of the abundance of work in this area of family connectedness, scholars of economics, sociology, education, and psychology now largely agree that parental education, socioeconomic status, and family structure have main effects on the academic performance of youth (Reardon, 2011) and an aggregated effect on system-level school performance indicators,

such as state-mandated standardized test scores (Bryk, Sebring, Allensworth, Luppescu, & Easton, 2010). In other words, students who experience more family related advantages and resources at home tend to perform better academically. Family support and involvement in schools have also been correlated with high student academic performance (Benard 2004; Christenson and Havsy 2004; Rumberger 1995). Displays of affection and a sense of feeling valued creates a safe space for a child to live in. Because of this connectedness, the child is able to thrive in other areas of development as well.

6.3.3 Ability to Make Meaning of Adversity (AMMA)

In the field of education, resilience studies provide evidence that many students succeed academically despite adverse economic conditions (Gamerzy, Masten, and Tellegen 1984; Rutter 1987; Benard 2004; Gizir and Aydin 2009), homelessness and transitory situations (Masten et al. 2008), violence and conflict-affected settings (Boyden 2003; Ungar 2005 and 2012), and social exclusion (Borma and Overman 2004). Resilience matters in education because learning and school success are not only possible in spite of adversity, education can also be the vehicle to overcome it. Individuals facing adversity seek to make sense of the situation they are experiencing and find a purpose that in turn will allow them to make meaningful and positive decisions. Since the 1970s many children who were raised in deprived, adverse environments actually went on to become successful and loving individuals (Luthar et al., 2000).

6.3.4 Family Communication and Problem Solving (FCPS)

Educators have recognized for many years that students' academic, social, and behavioural competence is associated with strong home support for learning. Recent research confirms that learning at home and outside of school differentiates high and low achievers in many schools

(Alexander, Entwisle, & Olson, 2001). Families that have learnt and applied the skills of effective communication in confronting issues that arise, and who find effective ways to solve problems, are able to create a climate conducive to learning and enhancing academic performance for learners within the household. Learners will thrive on the secure sense that he/she may pursue schoolwork, with the full knowledge the problems will be taken care of.

6.3.5 Utilising Social and Economic Resources (USER)

According to Luthar et al. (2000) a family's community and social support system can offer a rich, protective sense of belonging and cohesion. There is a growing appreciation for the broader view of resiliency as a complex interaction with community networks. Resilient families not only attain social support from their communities, but also give back to the community (Patterson, 2002). Extended kin and social networks can interact reciprocally with families to provide information, services, respite, avenues to contribute to the welfare of others, and companionship (Seccombe, 2002).

WESTERN CAPE

6.3.6 Family spirituality

Family resilience is fostered by shared beliefs that help members make meaning of crisis situations; facilitate a positive, hopeful outlook; and provide transcendent or spiritual values and purpose. Although religious coping is often depicted as passive, fatalistic, and acquiescent in recent research, Pargament (1997) has identified several types of religious coping strategies that are associated with different emotional sequelae. Religious coping functions in a variety of ways, such as through anxiety reduction, search for meaning, and social cohesiveness (Koenig et al., 2001). Further, religious scripture and belief provide behavioural guidelines for coping with problematic family relationships. Religious settings themselves provide models for religious coping behaviours as well as informal and formal sources of assistance, such as

clergy (Taylor, Ellison, Chatters, Levin, & Lincoln, 2000). Church-sponsored activities, such as family life education workshops and classes (e.g., marriage and parenting classes), provide novel information that individuals may use to evaluate problem events and resources for developing new coping skills and strategies to deal with problematic family interactions.

Recent research examining the nature of prayer and its role in individuals' coping with adversity indicates that prayer is a transformative personal experience that changes individuals in fundamental ways (e.g., forgiveness, stress reduction) and helps them to manage problems, whether the problems are life crises or daily hassles (Krause et al., 2000b; Taylor et al., 2004). As a religious coping behaviour, requesting prayer from others unambiguously signals the need for assistance from coreligionists and facilitates the provision of social support. Further, prayer from others reinforces social bonds and enhances group cohesion (Taylor et al., 2004). Finally, family prayer provides a model of coping behaviours and orientations, reinforces family cohesion and connectedness, and provides opportunities for the exchange of social support (e.g., emotional and spiritual support). Family prayer during stressful circumstances is an important ritual that provides information about the nature of the event (i.e., primary and secondary appraisals) and reinforces existing and emergent family role expectations (Howe, 2002).

Research shows that parental teaching, example, and dialogue about religious matters are important predictors of student academic performance. Religion plays a part in the coping process with respect to specific coping behaviours and strategies (e.g., spiritual support from others, prayer) as well as the enhancement and use of coping resources (Chatters, & Levin, 2004).

6.4 How children may compensate for difficulties they experience in the academic domain

What happens when children cannot claim that they feel academically competent? On the one hand, the findings of Roeser and his colleagues (Roeser & Eccles, in press; Roeser et al., 1995) suggest this is likely to lead to diminished self-esteem and social-emotional experience for many children. On the other hand, Roeser et al. (1995) also provides evidence of at least one mechanism of compensation: Youth who feel less academically competent and who receive lower marks also come to devalue academics the most. Another mechanism of compensation children can use is to change their valuing of, and engagement in achievement-related activities in other relevant life domains. Although there are several 'culturally mandated' dimensions of self that are likely central to esteem, including academics (eg. Stein, Markus & Roeser, 1995), some children may be able to compensate for difficulties in one area by achievement in another. For instance, in her expectancy-value formulation of achievement motivation, Eccles et al. (1983) suggests that competence and valuing of any number of achievement domains, including the performing and fine arts, sports, and the social sphere, can serve as relevant sources of esteem. Furthermore, in this model, self-perceptions of relative incompetence in one area are assumed to relate to children's feelings of competence and value in some other domain: those children with very low academic competence beliefs may develop other strengths and competencies to compensate for their relative lack of success at school. If successful, this strategy can provide the children with another ecological niche in which to develop a sense of self efficacy.

6.5 Limitations

No study is without its limitations. For this study the limitations were as follows:

- The study was conducted using self-reported questionnaires that might have indicated a measure of truth but not the actual truth as middle childhood participants were required to indicate *their own perceptions* of family resilience as *they* understood it. Other role players were not involved to compare these findings in a holistic manner.
- The study sample used, did not allow for generalization of these finding across all racial groups, as the majority of the study population identified themselves as 'coloured' and only a few participants identified themselves as 'other'. A convenient sampling method was employed, and this included only participants who are readily available and who agreed to participate in the study.

The original Family Resilience Assessment Scale was developed for adults. Although a pilot study (which was basically cognitive testing to check understanding of the questionnaire) was administered, one cannot be completely sure of the depth of understanding of the children.

6.6 Recommendations

Clinicians and practitioners have been ahead of researchers in conceptualizing family resilience (Compton & Hoffman, 2012). It would behove family scholars to consult some of the clinical literature on family resilience (see *Family Relations*, April 2002 for a Special Collection of more applied articles on family resilience). Family prevention science has documented the important role of parenting and family processes for child well-being and has identified specific family-level interactions as mediators of children's ability to adapt and thrive in the context of adversity. Interventions that include specific developmental guidance and psycho-education, as well as the opportunity to build and practice skills that support positive parenting practices,

parent-child relationships, and individual and family coping have been shown to enhance behavioural and emotional regulation in children (O'Connell et al., 2009). Moreover, family and relationships researchers have the benefit of learning from years of pioneering work on resilience conducted by psychiatrists and developmental psychologists. Often this work has clear relational and familial components (Masten, 2001). Family resilience is a natural extension of child resilience. This is an important aspect in the life of the middle childhood learner and knowledge of available literature will be a guide to whichever stakeholders are involved in the life of a child, resulting in a positive outcome.

Although a problem orientation predominates in family research, researchers facilitate developing a much more complex picture of family life. Families need assistance in accessing resources as well. Not all families deliberately isolate themselves, sometimes they are not sure of where to turn. There is no magic bullet for helping families overcome the effects either of major risks or minor daily risks but a better understanding of (a strengths orientation) resilience can help (Ganong, 2002).

The study focused on the relationship between family resilience and academic performance of learners in the phase of middle childhood. A healthy family environment sets a positive foundation for adulthood. As the results of this study suggest, when the process of family resilience is enhanced, the academic performance of learners is boosted. The study does highlight the need for further research particularly into family research.

A school learning community puts a laser-like focus on student learning and success (Epstein et al., 2002). School principals may start by introducing the topic in school assemblies, at school functions and in parent meetings, as well as educators promoting the concept within the classroom, which may then filter through as possible lessons. Although younger learners may not be able to comprehend the concept as such, and as it is gradually implemented into Life

Skills lessons at a basic level, and further expanded upon, the concept may become clearer.

This may lead to family discussions and enlightenment for whole families.

6.7 Conclusion

During middle childhood children have a need to develop competency, particularly in domains that are societally valued such as academic achievement. Students who experience more family related advantages and resources at home tend to perform better academically, and schools that enrol higher proportions of these students tend to be deemed successful by state accountability standards. Family strengths or resiliencies can be employed to resolve a child's problem. Findings of this study indicate that even in the midst of economical, physical and social challenges, when family resilience is prevalent, the academic development of the child is strengthened. The researcher trusts that the results of this study would act as a guide and framework for effective programs that could be implemented for the academic benefit of learners in middle childhood and for the enlightenment of families, specifically with regard to implementing family resilience.

REFERENCES

- Ahearn, E. M. (2000). *Increasing the participation of special needs students in NAEP*.

 Synthesis brief. Alexandria, VA: National Association of State Directors of Special Education.
- Alexander, K. L., Entwisle, D. R., & Horsey, C. S. (1997). From first grade forward: Early foundations of high school dropout. *Sociology of Education*, 70, 87–107.
- Alexander, K. L., Entwisle, D. R., & Kabbani, N. S. (2001). The drop-out process in life-course perspective: Early risk factors at home and school. *Teachers College Record*, 103, 760–822. http://dx.doi.org/10.1097/00004583-199007000-00014
- Alexander, K.L., Entwisle, D.R., & Olson, L.S. (2001). Schools, achievement, and inequality:

 A seasonal perspective. *Educational Evaluation and Policy Analysis*, 23(2), 171-191.
- Alvord, M. K., & Grados, J. J. (2005). Enhancing resilience in children: A proactive approach. *Professional Psychology: Research and Practice*, 36(3), 238–245, http://dx.doi.org/10.1037/0735-7028.36.3.238
- Amatea, E.S., Smith-Adcock, S. & Villares, E. (2006). *Professional School Counselling*, Vol. 9, No. 3, pp. 177-189.
- Anastasi, Anne. (1982). Psychological testing (5th ed.). New York: Macmillan.
- Anthony, E. J., & Cohler, B. J. (1987). The invulnerable child. New York: Guilford Press.
- Antonovsky, A., & Sourani, T. (1988). Family sense of coherence and family adaptation. *Journal of marriage and the family*, 50, 79-92.
- Aronowitz, T., & Morrison-Beedy, D. (2004). Resilience to risk-taking behaviours impoverished African-American girls: The role of mother-daughter connectedness.

 *Research in Nursing & Health, 27, 29-39.

- Arsenio, W.F., Cooperman, S., & Lover, A. (2000). Affective predictors of pre-schoolers' aggression and acceptance: Direct and indirect effects.
- Arthur, J., Waring, M., Coe, R. & Hedges, L.V. (2012) Research methods & methodologies in Education: Sage publications, 370.721 RES.
- Babbie, E. & Mouton, J. (2010). The practice of social research. Cape Town: Oxford University Press.
- Baker, T.L. (1994), Doing Social research (2nd Edn.), New York: McGraw-Hill Inc.
- Bastian, H. G. (2000). Musik(erziehung) und ihre Wirkung. Eine Langzeitstudie an Berliner Grundschulen. Mainz: Schott Musik International.
- Battle, J., & Lewis, M. (2002). The increasing significance of class: The relative effects of race and socioeconomic status on academic achievement. *Journal of Poverty*, 6(2), 21-35.
- Baumrind, D., (1991). The influence of parenting style on adolescent competence and substance use. *Journal of Early Adolescence*, 11(1), 56-95.

- Benard, B. (2004). Resiliency: What We have Learned. San Francisco: WestEd.
- Bermudez, J. M. (2013). Familias fuertes: *Family resilience among Latinos*. Springer New York. doi:10.1007/978-1-4614-3917-2_13
- Best, J. R., Miller, P. H., & Jones, L. L. (2009). Executive functions after age 5: Changes and correlates. *Developmental Review*, 29, 180–200. doi:10.1016/j.dr.2009.05.002
- Bhana, A., & Bachoo, S. (2011). The determinants of family resilience among families in low- and middle-income contexts: a systematic literature review. *South African Journal of Psychology*, 41(2), 2011, pp. 131-139 ISSN 0081-2463.

- Bjorklund, D. F. (2011). *Children's thinking: Cognitive development and individual differences* (5th ed.). Belmont, CA: Wadsworth
- Black, K., and M. Lobo (2008). "A conceptual review of family resilience factors," *Journal of Family Nursing*, 14(1), 33–55. Boss, P., *Family Stress Management: A Contextual Approach*, Newbury Park, CA: Sage, 2001.
- Blackorby, J., Chorost, M., Garza, N., & Guzman A.M. (2003). The academic performance of secondary school students with disabilities. In M. Wagner, C. Marder, J. Blackorby, R. Cameto, L. Newman,, P. Levine, & E. Davies-Mercier. *The achievements of youth with disabilities during secondary school*. Menlo Park, CA: SRI International.
- Bogin, B. (1997). Evolutionary hypotheses for human childhood. *Yearbook of Physical Anthropology*, 40, 63–89. doi:10.1002/(SICI)1096-8644
- Bonanno, G. A. (2004). Loss, trauma, and human resilience: Have we underestimated the human capacity to thrive after extremely aversive events? *American Psychologist*, 59, 20–28.
- Booth, A., & Amato, P. R. (2001). Parental Pre-divorce Relations and Offspring Post-divorce Well- Being. *Journal of Marriage and Family*, 63(1), 197-212.
- Bowlby, J. (1969/1982). Attachment and loss (Vol. 1, attachment). New York: Basic Books.
- Boyden, J. 2003. "Children under fire: Challenging assumptions about children's resilience." *Children, Youth and Environments 13*(1): 1-29.
- Bradbury, T. N., & Karney, B. R. (2004). Understanding and altering the longitudinal course of marriage. *Journal of Marriage and Family*, 66(4), 862-879. doi:10.1111/j.0022-2445.2004.00059.x

- Bronfenbrenner, U. (1986). Ecology of the family as a context for human development: Research perspectives. *Developmental Psychology*, 22, 723-742.
- Brooks, J. E. (2006). Strengthening resilience in children and youths: Maximizing opportunities in the schools. *Children and Schools*, 28(2), 69–76.
- Brooks-Gunn, J., Denner, J., and Klebanov, P.K. (1995). Families and neighbourhoods as contexts for education. In *Changing populations, changing schools: Ninety-fourth yearbook of the National Society for the Study of Education, Part II*. E. Flaxman and A. H. Passow, eds. Chicago, IL: National Society for the Study of Education, pp. 233–52.
- Bryk, A. S., Sebring, P. B., Allensworth, E. M., Luppescu, S., & Easton, J. Q. (2010). *Organizing schools for improvement: Lessons from Chicago*. Chicago, IL: University of Chicago Press.
- Cairns, R.B., Cairns, B.D., & Neckerman, H.J. (1989). Early School Dropout: Configurations and Determinants. *Child Development*, 60, 1437 1452.
- Campbell, B.C. (2011). Adrenarche and Middle Childhood. *Human Nature* 22 (3):327-349.
- Carneiro, P., & Heckman, J. (2003). *Human capital policy*. NBER working paper 9495. National Bureau of Economic Research: Cambridge, MA.
- Carnoy, M., Chisholm, L., et al., (2008). Towards Understanding Student Academic Performance in South Africa: A Pilot Study of Grade 6 Mathematics Lessons in South Africa. Pretoria: HSRC.
- Casas, F., Castellá Sarriera, J. Abs, D., Coenders, G., Alfaro, J., Saforcada, E., & Tonon, G. (2011). Subjective indicators of personal well-being among adolescents. Performance and results for different scales in Latin-language speaking countries: A contribution to the international debate. *Child IndicatorsResearch, online first.* doi:10.1007/s12187-011-9119-1.

- Chatters, L. M., Taylor, R. J., Lincoln, K. D., and Schroepfer, T. (2002). Patterns of informal support from family and church members among African Americans. *Journal of Black Studies*, *33*, 66–85.
- Cherlin, A.J., & Furstenberg, F.F. (1994). Stepfamilies in the United States: A reconsideration. *Annual Review of Sociology*, 20, 359-381. doi: 10.1146/annurev.so.20.080194.002043
- Cherniss, C., Extein, M., Goleman, D., & Weissberg, R.P. (2006). Emotional intelligence: What does the research really indicate? *Educational Psychologist*, 41(4), 239-245.
- Cherry, K. (2014) *Early Child Development*. Retrieved July 5, 2014, from psychology.about.com http://www//psychology.about.com
- Christenson, S., and L. Havsy. (2004). "Family-School-Peer Relationships: Significance for Social Emotional and Academic Support." In J. Zins, R. Weissberg, M. Wang, and H. Walberg. *Building academic success on social and emotional learning: What does the research say?* New York: Teachers College Press Columbia University.
- Cicchetti, D., & Rogosch, F.A. (1997). The role of self-organization in the promotion of resilience in maltreated children. *Development and Psychopathology*, *9*, 799-817.
- Clay, R. A. (2002). Research on 9/11: What psychologists have learned so far. Monitor on Psychology, 33, 28-30.
- Coleman, J. S., Campbell, E. Q., Hobson, C. J., McPartland, J., Mood, A. M., Weinfeld, F.
 D., & York, R. L. (1966). *Equality of educational opportunity*. Washington, DC: U.S.
 Department of Health, Education, and Welfare, Office of Education.
- Compton, W., & Hoffman, E. (2012). *Positive psychology: The science of happiness and flourishing*. Cengage Learning.
- Creswell, J. W., & Garrett, A. L. (2008). The" movement" of mixed methods research and the role of educators. *South African Journal of Education*, 28(3), 321-333.

- Creswell, J.W. (2009). *Research Design: Qualitative, Quantitative & mixed methods.* 3rd edition. Los Angeles. Sage Publications.
- Crosnoe, R., Johnson, M. K., & Elder, G. H. (2004). School size and the interpersonal side of education: An examination of race/ethnicity and organizational context. *Social Science Ouarterly*, 85(5), 12591274.
- Crouter, A.C., and A. Booth. (2003). *Children's Influence on Family Dynamics: The Neglected Side of Family Relationships*. Mahwah, NJ: Lawrence Erlbaum.
- Crouter, A. C., & McHale, S.M. (2005). Work, family, and children's time: implications for youth. In In S. M. Bianchi, L. M. Casper, and R.B. King (Eds.), *Work, Family, Health, and Well-Being*, (pp. 49-66). Mahwah: NJ: Lawrence Erlbaum and Associates.
- David, C. (2010). Quantitative Psychological Research. The complete students' companion, 3rd Edition: Psychology Department. Staffordshire University: Psychology Press.
- Del Giudice, M., Angeleri, R., & Manera, V. (2009). The juvenile transition: A developmental switch point in human life history. *DevelopmentalReview*, 29, 1–31. doi:10.1016/j.dr.2008.09.001
- Del Giudice, M. (2014). Middle childhood: An evolutionary developmental synthesis. Child Development Perspectives. doi:10.1111/cdep.12084

- Denham, S.A. (2006). Emotional competence. Implications for social functioning. In J. L. Luby (Ed.), *Handbook of preschool mental health. Development, Disorders and Treatment*, (pp.23-45). New York: The Guilford Press.
- Department of Social Development (2012). *White paper on families*. Pretoria: Department of Social Development.
- De Vaus, D.A. (1993), Surveys in Social Research (3rd edn.), London: UCL Press.
- DiPerna, J.C., & Elliot, S.N. (2000). *The academic competence evaluation scales* (ACES college). San Antonio, TX: The Psychological Association.

- Dishion, T. J., Shaw, D. S., Connell, A., Gardner, F., Weaver, C., & Wilson, M. (2008). The family check-up with high-risk indigent families: Outcomes of positive parenting and problem-behaviour from ages 2 through 4. *Child Development* (in press).
- Donaldson, S. J., & Ronan, K. R. (2006). The effects of sports participation on young adolescents' emotional well-being. *Adolescence*, 41, 369–389.
- Dryfoos, J.G. (1994). Full Service Schools: A revolution in health and social services for children, youth and families. San Francisco: Jossey-Bass Publishers.
- Duke, N. N., Borowsky, I. W., Pettingell, S. L., & McMorris, B. J. (2011). Examining youth hopelessness as an independent risk correlate for adolescent delinquency and violence. *Maternal and Child Health Journal*, 15(1), 87e97
- Durlak, J. A. (1997). Successful prevention programs for children and adolescents. Springer Science & Business Media.
- Durlak, J. A., Weissberg, R. P., Dymnicki, A. B., Taylor, R. D., & Schellinger, K. B. (2011). The impact of enhancing students' social and emotional learning: a meta-analysis of school-based universal interventions. *Child Development*, 82, 405–432. doi:10.1111/j.1467-8624.2010.01564.x.
- Durrant, J. (2007). Positive discipline, what it is and how to do it. Sweden: Save the Children.
- Eccles (*Parsons*), J., Adler, T.F., Futterman, R., Goff, S.B., Kaczala, C.M., Meece, J.L., & Midgley, C. (1983). Expectancies, values, and academic behaviours. In J.T. Spence (Ed.), *Achievement and achievement motivation* (pp. 75 146). San Francisco, CA.: W.H. Freeman.
- Eccles, J. S. (1999). The development of children ages 6 to 14. The future of children, 30-44.
- Edmonds, W. A., & Kennedy, T. D. (2010). A reference guide to basic research design for education and the social and behavioral sciences. New York, NY: Pearson

- Eisenberg, M. E., Olson, R. E., Neumark-Sztainer, D., Story, M., & Bearinger, L. H. (2004). Correlations between family meals and psychosocial well-being among adolescents. *Archives of Pediatric and Adolescent Medicine*, 158, 792–6.
- Elliott, S. N. (1998). Performance assessment of students' achievement: Research and practice. *Learning Disabilities Research and Practice*, 13(4), 233-241.
- Elliot S.N. &DiPerna, J. C. (2002). Promoting academic enablers to improve student achievement: An introduction to the mini-series. *School Psychology Review*, 31(3), 293.
- Epstein, N., Bishop, D., Ryan, C., Miller, I. and Keitner, G. (1993). The McMaster Model: view of healthy family functioning. In F. Walsh (ed.). Normal Family Processes (2nd edn, pp. 138-160). New York: Guilford Press.
- Epstein, J.L., (1995). School/family/community partnerships: Caring for the children we share. *Phi Delta Kappan*. 76(9), 701·712.
- Epstein, J. L., Sanders, M. G., Simon, B. S., Salinas, K. C., Jansorn, N. R., & Van Voorhis, F. L. (2002). *School, family, and community partnerships: Your handbook for action* (2nd ed.). Thousand Oaks, CA: Corwin.
- Epstein, N.B., Ryan, C.E, Bishop, D.S., Miller, I.W., & Keitner, G.I. (2003). The McMaster Model: A view of healthy family functioning. In F. Walsh (Ed.), *Normal family processes. Growing diversity and complexity* (3rd ed.; pp. 581–607). New York: Guildford.
- Espelage, E. (2014). Preventing youth violence and bullying through social-emotional school-based prevention programs and frameworks, in Donnelly, P.D. and Ward, C.

- L. Oxford Textbook of Violence Prevention: epidemiology, evidence, and policy, Oxford: Oxford University Press, forthcoming 6 November 2014.
- Feinberg, M. E., Kan, M. L., & Hetherington, E. M. (2007). Longitudinal study of coparenting conflict on adolescent maladjustment. *Journal of Marriage and Family*, 69, 687-702.
- Fiese, B. H., Tomcho, T. J., Douglas, M., Josephs, K., Poltrock, S., & Baker, T. (2002). A review of 50 years of research on naturally-occurring family routines and rituals:
 Cause for celebration? *Journal of Family Psychology*, 16(4), 381-390.
- Fink, A. (1995). The Survey Handbook, Sage Publication, Inc.
- Finley, G. E., Mira, S. D., & Schwartz, S. J. (2008). Perceived paternal and maternal involvement: Factor structures, mean differences, and parental roles. *Fathering*, 6, 62-82.
- Frain, M. P., Berven, N. L., Chan, F., & Tschopp, M. K. (2008). Family resiliency, uncertainty, optimism, and the quality of life of individuals with HIV/AIDS.

 *Rehabilitation Counseling Bulletin, 52, 16–27.

- Frude, N. (1991). *Understanding family problems. A psychological approach*. Chichester: John Wiley & Sons.
- Gamerzy, N., Masten, A.S., and Tellegen, A. (1984). "The study of stress and competence in children: a building block for developmental psychopathology." *Child Development* 55(1): 97-111.

- Ganong, L. H., & Coleman, M. (2002). Introduction to the Special Section: Family Resilience in Multiple Contexts. *Journal of Marriage and the Family*, 346-348.
- Gardner, F, Shaw, D.S, Dishion, T, Burton, J, Supplee, L. (2007). Randomized prevention trial for early conduct problems: Effects on proactive parenting and links to toddler disruptive behaviour. *Journal of Family Psychology*; 21:398–406. doi:10.1037/0893-3200.21.3.398.
- Gibson-Davis, C. (2008). The associations between complex family structures and cognitive and behavioral outcomes in middle childhood (Working paper). Durham, NC: Duke University.
- Giedd, J. N., & Rapoport, J. L. (2010). StructuralMRI of pediatric braindevelopment: What have we learned and where are we going? *Neuron*,67, 728–734. doi:10.1016/j.neuron.2010.08.040
- Gillespie, J.F.& Primaveray, J. (2000). Research on Family Resilience. *Journal of Prevention & Intervention in the Community*, 20:1-2, 1-3
- Gillespie, J.F. & Primaveray, J. (2014). *Diverse families, competent families: innovations in research and preventive intervention practice*. Routledge.
- Gizir, C.A., and Aydin, G. (2009). "Protective factors contributing to the academic resilience of students living in poverty in Turkey." *Professional School Counselling* 13(1): 38-49.
- Goldenberg, H., and Goldenberg, I. (2011). Family Therapy: An Overview, 8th ed. (Belmont, CA: Brooks/ Cole, 2012); Froma Walsh, *Strengthening Family Resilience*, 2nd ed. (New York: Guilford).

- Gottman, J. M. (2011). The Science of Trust: Emotional Attunement for Couples. New York:

 Norton
- Gray, D.E. (2009). Doing Research in the Real World, 2nd Edition. Sage Publications.
- Greene, B. A., & Miller, R. B. (1993, April). *Influences on course performance: Goals, perceived ability, and self-regulation*. Paper presented as a poster at the American Educational Research Association Annual Meeting, Atlanta, GA.
- Guilfoyle, S., Goebel, J., & Pai, A.L.H. (2011). Efficacy and flexibility impact perceived adherence barriers in paediatric kidney post-transplantation. *Families Systems & Health*, 29(1), 44-54. doi:10.1037/a0023024
- Han, W. (2008). The academic trajectories of children of immigrants and their school environments. *Developmental Psychology*, 44(6), 1572-1590.
- Hanson, S. M. (2001a). Family assessment and intervention. In S. M. Hanson (Ed.), Family health care nursing: *Theory, practice, and research* (2nd ed., pp. 170-195). Philadelphia: F.A. Davis.
- Hawley, D.R. and DeHaan, L. (1996). Family Process. Toward a Definition of Family Resilience: Integrating Life-Span and Family Perspectives. *Ph.D.2Volume 35*, Issue 3, pages 283–298.
- Henderson, V.L., & Dweck, C.S. (ref1990). Motivation and achievement. In S. S. Feldman & G. R. Etliott (Eds.). *At the threshold. The developing adolescent* (pp. 308-329). Cambridge, MA: Harvard University Press.
- Henry B., Caspi, A., Moffitt, T.E. and Silva, P.A. (1996). Temperamental and familial predictors of violent and nonviolent criminal convictions: age 3 to age 18, *Developmental psychology* 32(4), 614-623.

- Hetherington, M.E. (1989). "Coping with Family Transitions." Child Development, 60: 1-14.
- Hetherington, E.M. and Clingempeel, W.G. (1992) *Coping with Marital Transitions*.

 Chicago: Society for Research in Child Development.
- Hetherington, E.M., & Jodl, K. (1994). Stepfamilies as settings for child development. In A. Booth & J. Dunn (eds.), *Stepfamilies: Who benefits? Who does not?* (pp. 55-79). Hillsdale NJ: Erlbaum.
- Hetherington, E. M., Bridges, M., & Insabella, G. M. (1998). What matters? What does not? Five perspectives on the association between marital transitions and children's adjustment. *American Psychologist*, *53*(2), 167.
- Heubert, J., & Hauser, R. (1999). *High stakes: Testing for tracking, promotion, and graduation.*Washington, DC: National Research Council
- Higgins, E.T., and Parsons, J.E. (1983). Social cognition and the social life of the child:
 Stages as sub-cultures. In *Social cognition and social development*. E.T. Higgins,
 D.N. Ruble, and W.W. Hartup, eds. Cambridge, MA: Cambridge University Press,
 pp. 15-62.
- Hill, M. S. (1988). Marital stability and spouses shared time: A multidisciplinary hypothesis. *Journal of Family Issues*, 9(4), 427-451.
- Hochberg, Z. (2008). Juvenility in the context of life history theory. *Archives of Disease in Childhood*, 93, 534–539. doi:10.1136/adc.2008.137570
- Howe, G. W. (2002). Integrating family routines and rituals with other family research paradigms: Comment on the special section. *Journal of Family Psychology*, *16*, 437–440.
- Howes, P., & Markman, H. J. (1989). Marital quality and child functioning: A longitudinal investigation. *Child development*, 1044-1051.

- Jambon, M., & Smetana, J. G. (2014). Moral complexity in middle childhood: Children's evaluations of necessary harm. *DevelopmentalPsychology*, 50, 22–33. doi:10.1037/a0032992
- Jenkins, J.H. (1997). Not Without a Trace: Resilience and Remembering AmongBosnian Refugees (commentary on "A Family Survives Genocide"). Psychiatry, 60,40-43.
- Jensen, A. & Freedman, J. (2006). Assessment of School Climate. San Francisco: Six Seconds.
- Johnson, D. R., & Sharpe, M. N. (2000). Results of a national survey on the implementation of transition service requirements of IDEA. *Journal of Special Education Leadership*, 13, 15-26.
- Jonker, L., & Greeff, A. P. (2009). Resilience factors in families living with people with mental illnesses. *Journal of Community Psychology*, *37*, 859-873.
- Kaplan, H., Hill, K., Lancaster, J., & Hurtado, A. M. (2000). A theory of human life history evolution: Diet, intelligence, and longevity. *EvolutionaryAnthropology*, *9*, 156–185. doi:10.1002/1520-6505
- Kaufman, J., & Ziegler, E. (1987). Do abused children become abusive parents? *American Journal of Orthopsychiatry*, 57, 186-192.
- Kaufman, J., Cook, A., Arny, L., Jones, B., & Pittinsky, T. (1994). Problems defining resiliency: Illustrations from the study of maltreated children. *Development and Psychopathology*, 6, 215-229.

- King, V. & Sobolewski, J.M. (2006). Nonresident fathers contributions to adolescent well-being. *Journal of Marriage and Family*, 68, 537-557.
- Koenig, H. G., McCullough, M. E. & Larson, D. B. (2001). *Handbook of religion and health*.

 New York: Oxford University Press
- Koretz, D., & Hamilton, L. (1999). Assessing students with disabilities in Kentucky: The effects of accommodations, format, and subject. (CSE Technical Report No. CSE-TR-498). Los Angeles: University of California.
- Kowaleski-Jones, L., & Duncan, G.J. (1999). The structure of achievement and behaviour across middle childhood. *Child Development*, 4, 930-943.
- Krause, N., Chatters, L. M., Meltzer, T., and Morgan, D. L. (2000b). Using focus groups to explore the nature of prayer in late life. *Journal of Aging Studies*, *14*, 191–212.
- Kubey, R., & Csikszentmihalyi, M. (2002). Television addiction is no mere metaphor. Scientific American, 286(2), 74-80.
- Kuncel, N. R., Campbell, J. P., Hezlett, S. A., & Ones, D. S. (2001). Performance in college: The criterion problem. In D. S. Ones & S. A. Hezlett (Chairs), *Predicting performance: The interface of I/O psychology and educational research*. Symposium presented at the annual conference of the Society for Industrial and Organizational Psychology, San Diego, CA.
- Kuzawa, C. W., Chugani, H. T., Grossman, L. I., Lipovich, L., Muzik, O., Hof, P. R., . . . & Lange, N. (in press). Metabolic costs and evolutionary implications of human brain development. *Proceedings of the National Academy of Sciences of the United States of America*.

- Lagattuta, K. H., Sayfan, L., & Blattman, A. J. (2009). Forgetting commonground: Six- to seven-year-olds have an over interpretive theoryof mind. *Developmental Psychology*, 46, 1417–1432. doi:10.1037/a0021062
- Landau, S., Wykes, T., Reeder, C., Everitt, B., Knapp, M., Patel, A., Romeo, R. (2007).

 Cognitive remediation therapy in schizophrenia. Randomised controlled trial. *The British Journal of Psychiatry*, 190 (5) 421427; **DOI:** 10.1192/bjp.bp.106.026575
- Lareau, A. (2002). Invisible inequality: Social class and childrearing in black families and white families. *American Sociological Review*, 67(5), 747-776.
- Lebel, C., Walker, L., Leemans, A., Phillips, L., & Beaulieu, C. (2008). Microstructural maturation of the human brain from childhood toadulthood. *NeuroImage*, 40, 1044–1055. doi:10.1016/j.neuroimage.2007.12.053
- Leon, K. (2003). Risk and protective factors in young children's adjustment to parental divorce: A review of the research. *Family Relations*, 52(3), 258-270
- Lester, B., McGrath, M., Garcia-Coll, C., Brem, F., Sullivan, M. Mattis, S. (1995).
 Relationship between risk and protective factors, developmental outcome, and the home environment at four years of age in term and preterm infants. In Fitzgerald, H.,
 Lester, B. & Zuckerman, B. (Eds.), *Children Of Poverty: Research, Health and Policy Issues* (pp. 197-226). New York: Garland Publishing.
- Lifton, R. J. (1993). The Protean self: Human resilience in an age of fragmentation.

 New York: Basic Books.

- Luthar, S. S., & Ziegler, E. (1991). Vulnerability and competence: A review of research on resilience in childhood. *American Journal of Orthopsychiatry*, 61, 6-22.
- Luthar, S. S. (1999). Poverty and children's adjustment. Thousand Oaks, CA: Sage.
- Luthar, S. S., Cicchetti, D., & Becker, B. (2000). The construct of resilience: A critical evaluation and guidelines for future work. *Child Development*, 71, 543-562.
- Luthar, S. S. (2006). Resilience in development: A synthesis of research across five decades. In D. Cicchetti & D. J. Cohen (Eds.), *Developmental Psychopathology: Risk, disorder, and adaptation* (pp. 740–795). New York: Wiley.
- Maccoby, E. E., Depner, C. E. & Mnookin, R. H. (1990). Coparenting in the second year after divorce. *Journal of Marriage and the Family*, 52, 141-155. doi:10.2307/352846
- Mactavish, J., & Schleien, S. (1998). Playing together growing together: Parents' perspectives on the benefits of family recreation in families that include children with a developmental disability. *Therapeutic Recreation Journal*, 32(3), 207-230
- Manning, W. D., Stewart, S. D., & Smock, P. J. (2003). The complexity of fathers' parenting responsibilities and involvement with non-resident children. *Journal of Family Issues*, 24(5), 645-667.
- Marquez, P., Gil-Olarte, Palomera Martin, P., &Brackett, M.A. (2006). Relating emotional intelligence to social competence and academic achievement in high school students, *Psicothema*, 18(1), 118-123.
- Marsh, H.W., Trautwein, U., Ludtke, O., Koller, O., & Baumert, J. (2006). Academic self-concept, interest, grades and standardized test scores: Reciprocal effects models of causal ordering. *Child Development*, 76, 397–416.
- Masten, A. S., & Coatsworth, J. D. (1998). The development of competence in favorable and

- unfavorable environments: lessons from research on successful children. *American Psychologist*, 53, 205–220.
- Masten, A.S. (2001). Ordinary magic: Research processes in development, *American*Psychologist; 56: 227 238 (PubMed).
- Masten, A. S., Herbers, J. E., Cutuli, J. J., & Lafavor, T. L. (2008). Promoting competence and resilience in the school context. *Professional School Counseling*, *12*, 76–84. doi:10.5330/PSC.n.2010-12.76
- McCormick, C. M., Kuo, S.I-C., Masten, A.S (2011). "Developmental Tasks across the Lifespan," in *The Handbook of Lifespan Development*, ed. Karen L. Fingerman et al. (New York: Springer), 117-40. 4.
- McCreary, L. L., & Dancy, B. L. (2004). Dimensions of family functioning: Perspectives of low-income African American single-parent families. *Journal of Marriage and Family*, 66(3), 690-701.
- McCubbin, H. I., & McCubbin, M. A. (1988). Typologies of resilient families: Emerging roles of social class and ethnicity. *Family Relations*, *37*, 247-254.
- McCubbin, M. A., & McCubbin, H. I. (1993). Families coping with illness: The resiliency model of family stress, adjustment, and adaptation. In C. B. Danielson, B. Hamel-Bissell, & P. Winstead-Fry (Eds.), *Families, health, & illness: Perspectives on coping and intervention* (pp. 21-61). St. Louis, MO: Mosby.
- McCubbin, H. I., & McCubbin, M. A. (1996). Resiliency in families: A conceptual model of family adjustment and adaptation in response to stress and crises. In H. I.

 McCubbin, A.I., Thompson, E.A., & McCubbin, M. A. (1996). Family assessment:

- resiliency, coping and adaptation–Inventories for research and practice (pp. 1-64). Madison: University of Wisconsin System.
- McCubbin, H.I., Thompson, E.A., Thompson, A.I., & Fromer, J. E. (Eds.) (1998). *Stress, coping, and health in families: Sense of coherence and resiliency*. Thousand Oaks, CA: Sage Publications.
- McCubbin, H. I., Balling, K., Possin, P., Frierdich, S., & Bryne, B. (2002). Family resiliency in childhood cancer. *Family Relations*, *51*, 103-111.
- McGrath, R.E. (2011). Quantitative Methods in Psychology.
- McGrew, K. S., et al. (1995). Why we can't say much about the status of students with disabilities during education reform (Synthesis Report 21). Minneapolis, MN:

 National Center on Educational Outcomes, University of Minnesota. Available at http://www.education.umn.edu/nceo/OnlinePubs/Synthesis21.html
- McPherson, M., Smith-Lovin, L., & Brashears, M. E. (2006). Social isolation in America:

 Changes in core discussion networks over two decades. *American Sociological Review*, 71(3), 353-375

- Mederer, H. J. (1999). Surviving the demise of a way of life: Stress and resilience in Northeastern commercial fishing families. In H. I. McCubbin, E. A. Thompson, A. I. Thompson, & J. A. Futrell (Eds.). *The dynamics of resilient families* (pp. 203-235). Thousand Oaks, CA: Sage.
- Mestre, J.M., Guil, R., Lopes, P., Salovey, P., & Gil-Olarte, P. (2006). Emotional intelligence and social and academic adaptation to school, *Psicothema*, 18,112-117.

- Mikulincer, M., & Shaver, P. R. (2007). Boosting attachment security to promote mental health, prosocial values, and inter-group tolerance. *Psychological Inquiry*, *18*(3), 139-156.
- Miley, K.K., O'Melia, M. & DuBois, B. (2009). Generalist social work practice: An 562empowering approach. 6th ed. Boston: Allyn & Bacon.Milkie, M. A., Mattingly,
- Miller, P. H. (2011). Theories of developmental psychology (5th Ed.). New York: Worth.
- Minnema, J. E., Thurlow, M. L., Bielinski, J., & Scott, J. K. (2001). Past and current research on out-of-level testing of students with disabilities. *Assessment for Effective Intervention*, 26(2), 49-55
- Minuchin, S. (1974). Families and family therapy. Cambridge, MA: Harvard University.
- Moore, K.A., Chalk, R., Scarpa, J., & Vandivere, S. (2002). Family strengths: often overlooked, but real. Retrieved, from http://www.childtrends.org/Files/FamilyStrengths.pdf on 4 September, 2014.
- Morris, A. S., Silk, J. S., Steinberg, L., Sessa, F. M., Avenevoli, S., & Essex, M. J. (2002). Temperamental vulnerability and negative parenting as interacting predictors of child adjustment. *Journal of Marriage and Family*, 64(2), 461-471.
- Myers, H., Taylor, S., Alvy, K., Arrinton, A. & Richardson, M. (1992). Parental and family predictors of behaviour problems in inner-city black children. *American Journal of Community Psychology*, 20(5), 557-576.
- Newberger, C.M. (1980). The cognitive structure of parenthood: Designing a descriptive measure. *New Directions for Child and Adolescent Development*, 1980 (7), 45-67.

- Nomaguchi, K. M., Bianchi, S. M., & Robinson, J. P. (2004). The time squeeze: Parental statuses and feelings about time with children. *Journal of Marriageand Family*, 66(3), 739-761.
- Norris, R., Carroll, D., & Cochrane, R. (1992). The effects of physical activity and exercise training on psychological stress and well-being in an adolescent population. *Journal of Psychosomatic Research*, 36, 55–65. doi:10.1016/0022-3999(92)90114-H.
- Oberle, E., Schonert-Reichl, K. A., & Thomson, K. (2010). Understanding the link between social and emotional well-being and peer relations in early adolescence: gender specific predictors of peer acceptance. *Journal of Youth and Adolescence*, *39*, 1330 –1342. doi:10.1007/s10964-009-9486-9.
- O'Connell, M. E., Boat, T., & Warner, K. E. (Eds.). (2009). Preventing Mental, Emotional, and Behavioral Disorders Among Young People:: Progress and Possibilities.

 National Academies Press.
- Ogilvie, C.A., Newman, E., Todd, L. and Peck, D. (2014) Attachment and violent offending: a meta-analysis, *Aggression and Violent Behaviour* 19, 322–339.
- O'Leary, V. (1998) Strength in the face of adversity: individual and social thriving. *Journal* of Social Issues, 54, 425–446.
- Olson, D. H. (2000). Circumplex model of marital and family systems. *Journal of Family Therapy*, 22, 144-167.
- Olson, D. H., & Gorall, D.M. (2003). Circumplex Model of marital and family systems. In F. Walsh (Ed.), *Normal Family Processes: Growing Diversity and Complexity* (3rd ed., pp. 514-544). New York, NY: Guilford Press.

- Openshaw, K.P. (2011). The relationship between family functioning, family resilience, and the quality of life among vocational rehabilitation clients, Utah State University.
- Orbuch, T. L., Perry, C., Chesler, M., Fritz, J., & Repetto, P. (2005). Parent-child relationships and quality of life: Resilience among childhood cancer survivors. *Family Relations*, 54, 171-183.
- Orthner, D. K., Jones-Sanpei, H., & Williamson, S. (2004). The resilience and strengths of low-income families. *Family Relations*, 53(2), 159-167.
- Pargament, K. I. (1997). *The psychology of religion and coping: Theory, research, practice*. New York: Guilford.
- Parker, J.D.A., Creque, R.E., Barnhart, D.L., Harris, J.I., Majeski, S.A., Wood, L.M., Bond, B.J., & Hogan, M.J. (2004). Academic achievement in high school, does emotional intelligence matter? *Personality and Individual Differences*, 37, 1321-1330.
- Parker, J.D.A., Hogan, M.J., Eastabrook, J.M., Oke, A., & Wood, L.M. (2006). Emotional intelligence and student retention: Predicting the successful transition from high school to university. *Personality and Individual Differences*, 41, 1329-1336.

- Patterson, J. (2002). Understanding resilience. *Journal of Clinical Psychology*, Volume 58, Issue 3, pages 233–246
- Pearce, R. R., & Lin, Z. (2005). Cultural capital and postsecondary educational attainment among White and Chinese Americans: An analysis of NELS 1988–2000. *Asian American Policy Review*, 4, 19–37.

- Petrides, K. V., Frederickson, N., & Furnham, A. (2004). The role of trait emotional intelligence in academic performance and deviant behaviour at school. *Personality and Individual Differences*, 36, 277–293.
- Petrides, K.V., Sangareau, Y., Furnham, A., & Frederickson, N. (2006). Trait emotional intelligence and children's peer relations at school. *Social Development*, 15(3), 537-547.
- Poirel, N., Simon, G., Cassotti, M., Leroux, G., Perchey, G., & Lanoe, C., Houdé, O. (2011). The shift from local to global visual processing in 6-year-old children is associated with grey matter loss. *PLoS ONE*, 6, e20879. doi:10.1371/journal.pone.0020879
- Polit, D.F., Beck, C.T. and Hungler, B.P. (2001), *Essentials of Nursing Research: Methods, Appraisal and Utilization*. 5th Ed., Philadelphia: Lippincott Williams & Wilkins.
- Polloway, E.A., Bursuck, W., Plante, L., Epstein, M.H., Javanthi, M., McConeghy, J., (1994). Classroom grading: A national survey of policies. *Remedial and Special Education (RASE)*, 15(3), 162-170.
- Posner, J.K., & Vandell, D. L. (1999). After-school activities and the development of low-income urban children: a longitudinal study. Developmental Psychology, 35, 868–879. doi:10.1037/0012-1649.35.3.868.
- Potter, D., Mashburn, A., Grissner, D. (2013). *The Family, neuroscience and academic skills:*An interdisciplinary account of social class gaps in children's test scores, Volume
 42, Issue 2, March 2013, pgs. 446 464.

- Pretorius, T.B. (1998). Fortitude and stress-resistance: Development and validation of the Fortitude Questionnaire (FORQ). Bellville: University of the Western Cape.
- Pruett, M. K., Williams, T. Y., Insabella, G., & Little, T. D. (2003). Family and legal indicators of child adjustment to divorce among families with young children. *Journal of Family Psychology*, 17(2), 169-180.
- Rak, C., & Patterson, L. (1996). Promoting resilience in at-risk children. *Journal of Counseling and Development*, 74(4), 368-373. Available: http://www.valdosta.peachnet.edu/~whuitt/psy702/files/resbuild.html retrieved on August 30, 2014.
- Rasbash, J., Jenkins, J., O'Connor, T. G., Tackett, J. & Reiss, D. (2011). A social relations model of observed family negativity and positivity using a genetically informative sample. *Journal of Personality and Social Psychology*, 100(3), 474-491. doi:10.1037/a0020931
- Reardon, Sean. 2011. "The Widening Academic Achievement Gap Between the Rich and the Poor: New Evidence and Possible Explanations." Pp 91-115 in Whither Opportunity?:

 Rising Inequality, Schools, and Children's Life Chances, edited by G. J. Duncan and R. J. Murnane. New York: Russell Sage Foundation.
- Reilly, R. R. (1976). Factors involved in graduate student performance. *American Education Research Journal*, 13, 125–138.
- Richardson, G. E. (2002). The metatheory of resilience and resiliency. *Journal of*

- Clinical Psychology, 58, 307-321. doi:10.1002/jclp.10020
- Roeser, R.W., Midgley, C., Urdan, T. (1995). The middle school psychological environment and early adolescents' self-appraisals and academic engagement: The mediating role of goals and belonging. Manuscript submitted for review.
- Rumberger., R.W. 1995. "Dropping out of middle school: A multilevel analysis of students and schools." *American Educational Research Journal* 32(3): 583-625.
- Russell, R. V. (1996). *Pastimes: The context of contemporary leisure*. Dubuque, IA: Brown & Benchmark.
- Rutter, M. (1985). Resilience in the face of adversity: Protective factors and resistance to psychiatric disorders. *The British Journal of Psychiatry*, *147*, 598–611.
- Rutter, M. (1987). Psychosocial resilience and protective mechanisms. *American Journal of Orthopsychiatry*, 57, 316-331.

- Rutter, M. (1990). Psychosocial resilience and protective mechanisms. In J. Rolf, A. S. Masten, D. Cicchetti, K. H.Nuechterlein, & S. Weintraub (Eds.), *Risk and protective factors in the development of psychopathology* (pp. 181–214). Cambridge, UK: Cambridge University Press.
- Rutter, M. (1993). Resilience: some conceptual considerations. *Journal of Adolescent Health*, 14, 626-631.
- Rutter, M. (1999). Resilience concepts and findings: Implications for family therapy. *Journal of Family Therapy*, 21(2), 119-144.

- Salkind, N. (2010). Cross-Sectional Design. In Neil J. Salkind (Ed.), *Encyclopaedia of Research Design*. (pp. 314-315). Thousand Oaks, CA: SAGE Publications, Inc. doi: http://dx.doi.org.ezproxy.uwc.ac.za/10.4135/9781412961288.n96
- Samadi, A. (2007). Relationship between perception of support, of family, teachers, friends, and self-efficacy and academic achievement in middle school students in Arak, Faculty of Psychology and Educational Sciences, Allameh Tabatabai. University.
- Sameroff, A.J., and Haith, M.M. eds. (1996). *The five to seven year shift: The age of reason and responsibility*. Chicago: University of Chicago Press.
- Sanders, M. R. (1999). Triple P-Positive Parenting Program: Towards an empirically validated multilevel parenting and family support strategy for the prevention of behaviour and emotional problems in children. *Clinical Child and Family Psychology Review*, 2, 71-90.
- Sanders, M. R., McGee, E., Loureiro, T., & Murphy, M. (2013). Peer-Assisted Supervision and Support (PASS): Peer support and self-regulation training to promote effective implementation of evidence-based parenting interventions (manuscript in preparation). Saxton, J. (2000). *Investment in education: Private and public returns*. Retrieved from http://www.house.gov/jec/educ.pdf on 7 March, 2015.
- Scaramella, L.V., Leve, L.D. (2004. Clarifying parent–child reciprocities during early childhood: The early childhood coercion model. *Clinical Child and Family Psychology Review*.7:89–107.doi:10.1023/B:CCFP.0000030287.13160.a3.

- Seccombe, K. (2002). "Beating the odds" versus "Changing the odds": Poverty, resilience, and family policy. *Journal of Marriage and Family*, 64(2), 384-394.
- Seif, A.A. (2001). Educational psychology. Tehran: Agah
- Selman, R.L. (1980). The growth of interpersonal understanding. New York: Academic Press.
- Shaw, D.S. & Bell, R.Q. (1993). Developmental theories of parental contributors to antisocial behaviour, *Journal of Abnormal Child Psychology*, 30, 355-364.
- Shaw, D. S., Bell, R. Q., & Gilliom, M. (2000). A truly early starter model of antisocial behaviour revisited. *Clinical Child and Family Psychology Review*, *3*, 155–172.
- Shaw, D. S., Gilliom, M., Ingoldsby, E. M., & Nagin, D. (2003). Trajectories leading to school-age conduct problems. *Developmental Psychology*, 39, 189–200.
- Shaw, D. S., & Gross, H. (2008). Early childhood and the development of delinquency: What we have learned from recent longitudinal research. In A. Lieberman (Ed.), *The long view of crime: A synthesis of longitudinal research* (pp. 79–127). New York: Springer
- Shonkoff, J., Phillips, D.A., eds. 2000. From Neurons to Neighbourhoods: The Science of

 Early Childhood Development. Washington, DC: National Academy Press.
- Silliman, B. (1994). *Rationale for resilient families' concept paper*. National Network for Family Resiliency.
- Simon, J., Murphy, J., & Smith, S. (2005). Understanding and fostering family resilience. Family Journal, 13, 427–436.

- Sixbey, M.T. (2005). Development of the family resilience assessment scale to identify family resilience constructs. Doctoral dissertation, University of Florida (UMI Document Reproduction Service UMI Number: 3204501).
- Smaldone, A., Honig, J. C.,&Byrne, M.W. (2007). Sleepless inAmerica: inadequate sleep andrelationships to health and well-being of our nation's children. *Pediatrics*, 119, S29–37.
- Southwick, S.M., Litz, B.T., Charney, D. (2011). Resilience and Mental Health: Challenges

 Across the Lifespanhttps://books.google.co.za/books?isbn=113949886X
- Staveteig, S., & Wigton, A. (2005). Racial and ethnic disparities: Key findings from the national survey of America's families. Retrieved June 20, 2006, from http://www.urban.org/url.cfm?ID=309308&renderforprint=1
- Stein, K.F., Markus, H.R., & Roeser, R.W. (1995). The socio-cultural shaping of self-esteem in American early adolescent girls and boys. Manuscript submitted for review.
- Stein, M.T., & Perrin, E.L. (1998). Guidance for effective discipline. *Pediatrics*, 101, 723-728.
- Steptoe, A., & Butler, N. (1996). Sports participation and emotional wellbeing in adolescents. Lancet, 347, 1789–92.
- Stevenson, H.W., Chen, C., & Lee, S.Y. (1993). Mathematics achievement of Chinese, Japanese, and American children: Ten years later. *Science*, 259, 53-58.
- Stinnett, N., & DeFrain, J. (1985). Secrets of strong families. Boston: Little, Brown.

- Stipek, D. & Gralinski, H. (1996). Children's theories of intelligence and school performance. *Journal of Educational Psychology*, 88, 711-723.
- Suarez-Orozco, C., Pimentel, A., Martin, M. (2009). The Significance of Relationships:

 Academic Engagement and Achievement among Newcomer Immigrant Youth.

 Teachers College Record Volume 111, Number 3, pp. 712–749.
- Sykes, WR. J Relig Health. (1989) Jun;28(2):97-108. doi: 10.1007/BF00996067.

 PMID: 24276815
- Taylor, R. J., Ellison, C. G., Chatters, L. M., Levin, J. S., and Lincoln, K. D. (2000). Mental health services in faith communities: The role of clergy in Black churches. *Social Work*, 45, 73–87.
- Taylor, R. J., Chatters, L. M. & Levin, J. S. (2004). Religion in the lives of African

 Americans: Social, psychological, and health perspectives. Thousand Oaks, CA:

 Sage.
- Teo, A.E., Carlson, P.J., Mathieu, B.E., Stroufe, L.A. (1996). A prospective longitudinal study of psychosocial predictors of achievement. *Journal of School Psychology*. 34:285-306.
- Theokas, C., Almerigi, J., Lerner, R. M., Dowling, E. M., Benson, P. L., & Scales, P. C. (2005). Conceptualizing and modelling individual and ecological asset components of thriving in early adolescence. Journal of Early Adolescence, 25, 113–143.
- To provide for a uniform system for the organisation. WCP (1977), Act No 12.

- Toth, J. F, Brown, R. B, & Xu, X., (2002). Separate family and community realities? An urban-rural comparison of the association between family life satisfaction and community satisfaction. *Community, Work and Family, 5*(2), 181-202.
- Tremblay, R. E., Boulerice, B., Arseneault, L., & Niscale, M. J. (1995). Does low self-control during childhood explain the association between delinquency and accidents in early adolescence? *Criminal Behaviour and Mental Health*, 5, 439–451.
- Trentacosta, C. J., Hyde, L. W., Shaw, D. S., Dishion, T. J., Gardner, F., & Wilson, M. (2008). The relations among cumulative risk, parenting, and behaviour problems during early childhood. *Journal of Child Psychology & Psychiatry*, 49, 1211-1219.
- Trinidad, D. R., & Johnson, C. A (2002). The association between emotional intelligence and early adolescent tobacco and alcohol use, *Personality and Individual Differences*, 32, 95–105.
- Tusaie, K. & Dyer, J. (2004). Resilience: A historical review of the construct. *Holistic Nursing Practice*, 18(1), 3-8. doi:10.1097/00004650-200401000-00002
- Ungar, M. (ed.) 2005. Handbook for Working with Children and Youth: Pathways to Resilience Across Cultures and Contexts. Thousand Oaks: Sage Publications.
- Ungar, M. (ed.). 2012. The Social Ecology of Resilience: A Handbook of Theory and *Practice*. New York: Springer.
- Van der Berg, S. & Louw (2006), M. Lessons learnt from SACMEQII: South African student performance in regional context. University of Stellenbosch, Department of Economics.

- Ve´ronneau, M.-H., Vitaro, F., Pedersen, S., & Tremblay, R. E. (2008). Do peers contribute to the likelihood of secondary school graduation among disadvantaged boys? *Journal of Educational Psychology*, 100, 429–442.
- Voydanoff, P. (2004). The effects of work demands and resources on work-to-family conflict and facilitation. *Journal of Marriage and the Family*, 66, 398–412.
- Walsh, F. (1996). *The concept of Family Resilience*: Crisis and Challenge. Fam Proc 35: 261-281.
- Walsh, F. (1998). Strengthening family resilience. New York: The Guilford Press.
- Walsh, F. (Ed.). (1999). Spiritual resources in family therapy. New York: Guilford Press.
- Walsh, F. (2002). A family resilience framework: Innovative practice applications. *Family Relations*, 51, 130- 137.
- Walsh, F. (2003). Family resilience: A framework for clinical practice. *Family Process*, 42(1), 1-18.

UNIVERSITY of the

- Weisner, T. S. (1996). The 5–7 transition as an ecocultural project. InA. Sameroff & M. Haith (Eds.). *The five to seven year shift: The ageof reason and responsibility* (pp. 295–326). Chicago, IL: University of Chicago Press.
- Wells, J. C. K. (2007). Sexual dimorphism of body composition. *Best Practice & Research Clinical Endocrinology & Metabolism*, 21, 415–430. doi:10.1016/j.beem.2007.04.007
- Werner, E. E. (1993). Risk, resilience, and recovery: Perspectives from the Kauai longitudinal study. *Development and Psychopathology*, *5*, 503-515.

- Werner, E. (1995). Resilience in development. *Current Directions in Psychological Science*, 4, 81-85.
- Werner, E., & Smith, R. (1977). *Kauai's children come of age*. Honolulu: University ofHawaii Press.
- White, N., Richter, J., Koeckeritz, J., Munch, K., & Walter, P. (2004). "Going forward": Family resiliency in patients on hemodialysis. *Journal of Family Nursing*, 10(3), 357-378.
- Wolin, S. J., & Wolin, S. (1993). The resilient self. New York: Random House.
- Woolley, M. E., & Grogan-Kaylor, A. (2006). Protective family factors in the context of neighborhood: Promoting positive school outcomes. *Family Relations*, 55, 93-104.
- Wyman, P. A., Cowen, E. L., Work, W. C., & Kerley, J. H. (1993). The role of children's future expectations in self-system functioning and adjustment to life stress: a prospective study of urban at-risk children. *Development and Psychopathology*, 5, 649e661.
- Zabriskie, R. B., & McCormick, B. P. (2001). The Influences of Family Leisure Patterns on Perceptions of Family Functioning. Family Relations, 50(3), 281-289.
- Zimmerman, M.A. and Arunkumar, R. (1994). Social Policy Report. Society for Research in Child Development. *Resiliency Research: Implications for Schools and Policy*, Volume VIII, Number 4.
- Zins, J., Weissberg, R.W., Wang, M.C., & Walberg, H. (2004). *Building school success on social emotional learning: What does the research say?* New York: Teachers College Press.



APPENDIX I

FAMILY RESILIENCE ASSESSMENT SCALE

Please read each statement carefully. Decide how well you believe it describes your family now from your viewpoint. Your family may include any individuals you wish.

	Strongly Agree	Agree	Disagree	Strongly Disagree
Our family structure is flexible to deal with the unexpected				
2. Our friends value us and who we are				
3. The things we do for each other make us feel a part of the family				
4. We accept stressful events as a part of life				
5. We accept that problems occur unexpectedly				
6. We all have input into major family decisions				
7. We are able to work through pain and come to an understanding				
8. We are adaptable to demands placed on us as a family				
9. We are open to new ways of doing things in our family				
10. We are understood by other family members IVERSITY of th	g			
11. We ask neighbors for help and assistance	3			
12. We attend church/synagogue/mosque services				
13. We believe we can handle our problems				
14. We can ask for clarification if we do not understand each other				
15. We can be honest and direct with each other in our family				
16. We can blow off steam at home without upsetting someone				
17. We can compromise when problems come up				
18. We can deal with family differences in accepting a loss				
19. We can depend upon people in this community				
20. We can question the meaning behind messages in our family				
21. We can solve major problems				
22. We can survive if another problem comes up				
23. We can talk about the way we communicate in our family				

24. We can work through difficulties as a family			
25. We consult with each other about decisions			
26. We define problems positively to solve them			
27. We discuss problems and feel good about the solutions			
28. We discuss things until we reach a resolution			
29. We feel free to express our opinions			
30. We feel good giving time and energy to our family			
31. We feel people in this community are willing to help in an emergency			
32. We feel secure living in this community			
33. We feel taken for granted by family members			
34. We feel we are strong in facing big problems			
35. We have faith in a supreme being			
36. We have the strength to solve our problems			
37. We keep our feelings to ourselves			
38. We know there is community help if there is trouble			
39. We know we are important to our friends			
40. We learn from each other's mistakes			
41. We mean what we say to each other in our family	e		
42. We participate in church activities			
43. We receive gifts and favors from neighbors			
44. We seek advice from religious advisors			
45. We seldom listen to family members' concerns or problems			
46. We share responsibility in the family			
47. We show love and affection for family members			
48. We tell each other how much we care for one another			
49. We think this is a good community to raise children			
50. We think we should not get too involved with people in this community			
51. We trust things will work out even in difficult times			
52. We try new ways of working with problems			

53. We understand communication from other family members		
54. We work to make sure family members are not emotionally or physically hurt		



APPENDIX II

FAMILY RESILIENCE ASSESSMENT SCALE (REVISED)

Please read each statement carefully. Decide how well you believe it describes your family now from your viewpoint. Your family may include any individuals you wish.

		Strongly Agree	Agree	Disagree	Strongly Disagree
1.	Our family is flexible to deal with unexpected events eg. moving home.				
2.	Our friends value us and who we are				
3.	The things we do for each other make us feel a part of the family				
4.	We accept stressful events as a part of life				
5.	We accept that problems which we do not expect, can happen.				
6.	We all have input into major family decisions				
7.	We are able to work through pain and come to an understanding				
8.	We are flexible to cope with demands placed on us as a family				
9.	We are open to new ways of doing things in our family				
10.	We are understood by other family members				
11.	We ask neighbors for help and assistance				
12.	We attend church/synagogue/mosque services				
13.	We believe we can handle our problems				
14.	We can ask for an explanation if we do not understand each other				
15.	We can be honest and direct with each other in our family				
16.	We can show our feelings at home without upsetting someone				
17.	We can 'make a deal' with each other when problems come up				
18.	We can deal with family differences in accepting a loss				
19.	We can depend upon people in this community				
20.	We can question the meaning behind messages in our family				
21.	We can solve major problems				
22.	We can survive if another problem comes up				
23.	We can talk about the way we communicate in our family				

24. We can work through difficulties as a family		
25. We talk to each other about decisions we need to make		
26. We define problems positively to solve them		
27. We discuss problems and feel good about the solutions		
28. We discuss things until we reach a resolution		
29. We feel free to express our opinions		
30. We feel good giving time and energy to our family		
31. We feel people in this community are willing to help in an emergency		
32. We feel secure living in this community		
33. We feel taken advantage of by family members		
34. We feel we are strong in facing big problems		
35. We have faith in a supreme being		
36. We have the strength to solve our problems		
37. We keep our feelings to ourselves		
38. We know there is community help if there is trouble		
39. We know we are important to our friends		
40. We learn from each other's mistakes		
41. We mean what we say to each other in our family		
42. We take part in church/mosque activities		
43. We receive gifts and favors from neighbors		
44. We seek advice from religious advisors, such as priests, imams etc.		
45. We seldom listen to family members' concerns or problems		
46. We share responsibility in the family		
47. We show love and affection for family members		
48. We tell each other how much we care for one another		
49. We think this is a good community to raise children		
50. We think we should not get too involved with people in this community		
51. We trust things will work out even in difficult times		
	<u> </u>	1

52. We try new ways of working with problems		
53. We understand how other family members talk to each other		
54. We work to make sure family members are not emotionally or physically hurt		



APPENDIX III



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2277/2970

E-mail: rehtse30@hotmail.com



Project Title: The relationship between family resilience and academic performance of learners in the phase of middle childhood.

This is a research project being conducted by Esther Alard (a master's student) at the University of the Western Cape. We are inviting you to participate in this research project because you are involved in the academic performance of the student who will be answering the questionnaire. The purpose of this research project is to examine the relationship between perceived family resilience (how families cope with problems) and the academic performance of children in the phase of middle childhood. The reason this knowledge is being sought is that families will be made aware of how to overcome certain barriers and motivate each other and thereby aid the academic performance of middle childhood learners.

You will be asked to complete a questionnaire in this study. Questions range from how families talk to one another to how they deal with problems. These questions will be answered by ticking one of the appropriate boxes: Strongly agree, agree, disagree and strongly disagree. The procedure will take place at the school and the duration of it will be approximately 30 minutes.

Would my participation in this study be kept confidential?

We will do our best to keep your personal information confidential. To help protect your confidentiality, participation in the questionnaire will be anonymous. The surveys will not contain information that may personally identify you. Completed questionnaires will be kept in secure storage areas and only the researcher will have access to them. If we write a report or article about this research project, your identity will be protected to the maximum extent possible.

There are no known risks associated with participating in this research project.

What are the benefits of this research? ERSITY of the

We hope that, in future, other people might benefit from this study through improved understanding of family resilience.

Your participation in this research is completely voluntary. (You may choose not to take part at all. If you decide to participate in this research, you may stop participating at any time. If you decide not to participate in this study or if you stop participating at any time, you will not be penalized or lose any benefits to which you otherwise qualify)

If you have any questions about the research study itself, please contact Esther Alard at: 076 826 9742, e-mail address: rehtse30@hotmail.com.

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Head of Department: Professor R. Schenk

Dean of the Faculty of Community and Health Sciences: Professor Joze Frantz

University of the Western Cape

Private Bag X17

Bellville 7535

This research has been approved by the University of the Western Cape's Senate Research

Committee and Ethics Commit

UNIVERSITY of the WESTERN CAPE

APPENDIX IV



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2277/2970

E-mail: rehtse30@hotmail.com

INFORMATION SHEET FOR PARENTS

Project Title: The relationship between family resilience and academic performance of learners in the phase of middle childhood.

This is a research project being conducted by Esther Alard (a master's student) at the University of the Western Cape. We are inviting you to participate in this research project because you are involved in the academic performance of the student who will be answering the questionnaire. The purpose of this research project isto examine the relationship between perceived family resilience and the academic performance of children in the phase of middle childhood. The reason this knowledge is being sought is that families will be made aware of how to overcome certain barriers and motivate each other and thereby aid the academic performance of middle childhood learners.

You will be asked to consent to your child's participation in the study. The study will take the form of a questionnaire, which consists of 54 questions.

Questions pertain to how families collaborate with each other and how they deal with problems. These questions will be answered by ticking one of the appropriate boxes: Strongly agree, agree, disagree and strongly disagree. The procedure will take place at the school and the duration of it will be approximately 30 minutes.

Would my child's participation in this study be kept confidential?

We will do our best to keep your child's personal information confidential. To help protect his/her confidentiality, participation in the questionnaire will be anonymous. The surveys will not contain information that may personally identify him/her. Completed questionnaires will be kept in secure storage areas and only the researcher will have access to them. If we write a report or article about this research project, your child's identity will be protected to the maximum extent possible.

There are no known risks associated with participating in this research project.

What are the benefits of this research?

We hope to contribute to the knowledge base of family psychology and, in future, other people might benefit from this study through improved understanding of family resilience.

WESTERN CAPE

Your child's participation in this research is completely voluntary. (He/She may choose not to take part at all. If your child decides to participate in this research, he/she may stop participating at any time. If your child decides not to participate in this study or if he/she stops participating at any time, your child will not be penalized or lose any benefits to which he/she otherwise qualifies).

If you have any questions about the research study itself, please contact Esther Alard at: 076 826 9742, e-mail address: rehtse30@hotmail.com.

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Head of Department: Professor R. Schenk

Dean of the Faculty of Community and Health Sciences: Professor Joze Frantz

University of the Western Cape

Private Bag X17

Bellville 7535



This research has been approved by the University of the Western Cape's Senate Research Committee and Ethics Commit



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2277/2970

E-mail:rehtse30@hotmail.com

CON	ISENT FORM FOR PARTICIPANTS
Title of Research Project:	
The relationship between fa	amily resilience and academic performance of learners in the
phase of middle childhood	UNIVERSITY of the WESTERN CAPE
The study has been described	to me in a language that I understand and I freely and voluntarily
agree to participate. My ques	tions about the study have been answered. I understand that my
identity will not be disclosed	and that I may withdraw from the study without giving a reason
at any time and this will not r	negatively affect me in any way.
Participant's name	
Participant's signature	•••••
Date	••••••

Should you have any questions regarding this study or wish to report any problems you have experienced related to the study, please contact the study coordinator:

Study Coordinator's Name: Dr N Roman

Private Bag X 17, Bellville 7535, South Africa

Tel: (021) 959-3095

E-mail:nroman@uwc.ac.za



APPENDIX VI



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa

Tel: +27 21-959 2277/2970

E-mail:rehtse30@hotmail.com

CONSENT FORM FOR PARENTS

Title of Research Project:



The relationship between family resilience and academic performance of learners in the phase of middle childhood

The study has been described to me in a language that I understand and I freely and voluntarily allow my child to participate. My questions about the study have been answered. I understand that my child's identity will not be disclosed and that he/she may withdraw from the study without giving a reason at any time and this will not negatively affect him/her in any way.

Parent's name	•••••
Parent's signature	
Date	

Should you have any questions regarding this study or wish to report any problems you have experienced related to the study, please contact the study coordinator:

Study Coordinator's Name: Dr N Roman

Private Bag X 17, Bellville 7535, South Africa

Tel: (021) 959-3095

E-mail:nroman@uwc.ac.za



APPENDIX VII

DEMOGRAPHIC INFORMATION

Please complete the following by marking the correct response with an X.

Gender	Male			Female			
Age							
Grade							
Race	Colou	ıred	Black/ African	Wh	ite	Indian /	/ Asian
Home language	Afrika	aans	English	IsiXhosa		Other	
I live with my(tick as many as you have in your house)	Mother	Father UNIV	Grandparent VERSITY of the TERN CAPE	Uncle/Aunt		Sibling	Child
Is there a child with a disability living in your home?			Yes			No	

FAMILY RESILIENCE QUESTIONNAIRE

This questionnaire contains a number of statements related to opinions and feelings about you and life in general. Read each statement carefully, then indicate the extent to which you agree or disagree by indicating one of the alternative categories provided. For example, if you **Strongly Agree**, indicate **SA**. If you are **Agree**, indicate **A**. If you **Strongly Disagree**, indicate **SD**. If you **Disagree**, indicate **D**.



- 1. Our past achievements have given our life meaning and purpose ______
- 2. In our life we have very clear goals and aims
- 3. We have discovered a satisfying life purpose